

Article A Competitive Newsvendor Problem with Product Substitution under the Carbon Cap-and-Trade System

Yuheng Ren¹, Wenliang Bian^{1,2}, Haicheng Li^{1,*} and Yingjie Ju¹

- ¹ School of Economics and Management, Beijing Jiaotong University, Beijing 100044, China; 19113045@bjtu.edu.cn (Y.R.); wlbian@bjtu.edu.cn (W.B.); 19113039@bjtu.edu.cn (Y.J.)
- ² Research Center for Central and Eastern Europe, Beijing Jiaotong University, Beijing 100044, China

* Correspondence: 22120597@bjtu.edu.cn

Abstract: This study investigates the competitive issues of the newsvendor problem with product substitution under the carbon cap-and-trade system. Building on existing research, this paper introduces the carbon cap-and-trade system under uncertain market demand and considers that the original equipment manufacturer (OEM) can choose to procure raw materials from the contract manufacturer (CM), with both final products being substitutable. Furthermore, we explore the different substitution relationships between OEM and CM products under both pure competitive and co-opetitive modes. For this problem, decision models are developed for various scenarios, and optimal solutions satisfying given conditions are provided. We find that in one-way substitution, under pure competition, an increase in the OEM's (or CM's) green investment and substitution rate only leads to an increase in OEM's (or CM's) yields, while an increase in the OEM's (or CM's) green investment does not necessarily reduce CM's (or OEM's) yields. In the co-opetitive mode, an increase in the substitution rate and green investments of both manufacturers may lead to an increase in the yields of both manufacturers. Furthermore, an increase in carbon trading prices does not necessarily inhibit the manufacturer's yields. Moreover, we find that under the same competition mode, under certain conditions, two-way substitution between OEM and CM can bring better profits to both manufacturers and the entire supply chain. When the two modes are in the same substitution scenario, and the CM cannot substitute for the OEM, the optimal decisions and total supply chain profits of the two modes are equal. Finally, through numerical analysis, we find that neither mode is necessarily optimal when CM can substitute for OEM. Additionally, it is observed that when the same mode is in different substitution scenarios, total supply chain profits may be enhanced in the presence of product substitution.

Keywords: carbon cap-and-trade system; competition; product substitution; low-carbon supply chain

1. Introduction

With the increase in global greenhouse gas emissions, significant damage to the global climate has been observed [1]. This damage is primarily caused by excessive carbon emissions in industrial production processes [2,3]. Consequently, to effectively reduce carbon dioxide emissions, many countries have signed the Kyoto Protocol and The Paris Agreement, and have implemented various carbon emission reduction policies [1]. As carbon emission reduction policies are successively implemented, carbon-emitting enterprises and their stakeholders are facing immense pressure to reduce emissions [4].

Carbon cap-and-trade systems are considered one of the most effective emission reduction policies [2,5–7], and have been adopted by several countries, such as the European Union Emission Trading Scheme (EU-ETS), the California Scheme and China Carbon Emission Trade Exchange (CCETE) [8]. Under the carbon cap-and-trade system, at the beginning of each period, emission-reducing enterprises are allocated a certain amount of carbon allowances. During this period, if a company's carbon emissions do not exceed



Citation: Ren, Y.; Bian, W.; Li, H.; Ju, Y. A Competitive Newsvendor Problem with Product Substitution under the Carbon Cap-and-Trade System. *Systems* **2024**, *12*, 201. https:// doi.org/10.3390/systems12060201

Academic Editor: Wen-Chyuan Chiang

Received: 14 May 2024 Revised: 3 June 2024 Accepted: 5 June 2024 Published: 8 June 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). its carbon allowance, it can sell the surplus allowances on the carbon trading market to generate additional revenue. Similarly, if a company's carbon emissions exceed its carbon allowance, it can purchase allowances on the carbon trading market to fulfill its carbon quota obligation [4,6,8]. As the carbon cap-and-trade system may bring additional revenue to emission-reducing enterprises, their incentive to reduce emissions can be significantly enhanced [6].

Although the carbon cap-and-trade system can effectively enhance companies' enthusiasm for emission reduction, they still face threats from other risks during the emission reduction process. Since all companies within the same industry are required to reduce emissions, competitors of emission-reducing enterprises may offer similar or substitutable green products [9]. This phenomenon can lead to significant financial risks for emission-reducing enterprises that invest heavily but are unable to sell enough products [10]. However, co-operation among emission-reducing enterprises is also possible. For example, China Petrochemical Corporation is responsible for both oil extraction and refining, as well as wholesale and retail of finished oil products. On the other hand, China National Aviation Fuel Group Limited is only responsible for the procurement and sales of aviation fuel. China Petrochemical Corporation is both a competitor and a supplier of China National Aviation Fuel Group Limited. There is both competition and co-operation between the two entities. Inspired by this example, we will explore how an original equipment manufacturer (OEM) and a contract manufacturer (CM), two manufacturers under the carbon cap-andtrade system, can respond to demand uncertainty through competition and co-operation to achieve better supply chain performance.

Furthermore, during co-operation, both manufacturers sell substitutable products. When one manufacturer experiences a shortage, customers will shift their purchases to the other manufacturer. In other words, when the OEM (or CM) is unable to satisfy all of the customers' demands, there will be a fraction of unfulfilled demand that will be satisfied by the CM (or OEM). If the OEM's (or CM's) customers shift their purchases to CM (or OEM) because the OEM (or CM) is out of stock, the OEM (or CM) suffers a certain amount of lost sales, while the CM (or OEM) can earn additional sales revenue. The degree of product substitution is related to customer purchase loyalty [11]. Purchase loyalty can lead to greater market demand but does not result in higher product prices [12]. In this study, we adopt the same assumption as [11], namely that the degree of product substitution is related to the level of customer purchase loyalty. When customers have low purchase loyalty towards a manufacturer, their products are more easily substituted. When purchase loyalty is at zero, one-way substitution occurs [11,13].

Currently, there is limited research on the competitive issues considering demand uncertainty under the carbon cap-and-trade system. Given this context, we will investigate the following questions. Firstly, how does product substitution affect the decisions of supply chain members under the carbon cap-and-trade system? Secondly, what impact do carbon trading prices and manufacturers' green investments have on supply chain decisions? Finally, under different substitution scenarios, which competitive mode selection can bring better profits to the supply chain?

To address this issue, we first focus on the research of the purely competitive mode. Under this mode, we construct a newsvendor model for OEM purchasing from a third-party supplier in scenarios of no substitution, one-way substitution, and two-way substitution. Then, for the co-opetitive mode, we construct a newsvendor model for OEM purchasing from the CM in scenarios of no substitution, one-way substitution, and two-way substitution. Finally, we compare the optimal decisions and profits under both modes for the same substitution scenario.

This study contributes to the research on the competitive issues of the newsvendor problem considering product substitution under the carbon cap-and-trade system. Firstly, by constructing a newsvendor model under different scenarios, we derive optimal decisions satisfying given conditions for various situations. Next, we analyze the impact of substitution rates, green investments, and carbon trading prices on optimal decisions under different scenarios. Finally, this paper explores how supply chain members under the carbon cap-and-trade system can achieve better supply chain performance through competition and co-operative. Based on these findings, we find that when CM's products cannot substitute for OEM's products, the optimal decisions and total supply chain profits are the same under both purely competitive and co-opetitive modes. However, when CM's products can substitute for OEM's products, the co-opetitive mode may not necessarily lead to better supply chain performance. Additionally, we find that an increase in carbon trading prices does not necessarily suppress manufacturer's yields.

The structure of this paper is as follows. Section 2 provides a literature review of relevant studies. Section 3 defines variables and describes the models. Section 4 analyzes the optimal decisions of the purely competitive mode under different substitution scenarios and examines the impact of substitution rates, green investments, and carbon trading prices on manufacturers' optimal decisions. Section 5 analyzes the optimal decisions of the co-opetitive mode under different substitution scenarios and investigates the impact of substitution rates, and carbon trading prices on manufacturers' optimal decisions. Section 5 analyzes the optimal decisions of the co-opetitive mode under different substitution scenarios and investigates the impact of substitution rates, green investments, and carbon trading prices on manufacturers' optimal ordering decisions. Section 6 presents a numerical analysis. Section 7 summarizes the conclusions.

2. Literature Review

With the increasing awareness of carbon emissions reduction worldwide, various carbon emission reduction policies have been introduced. Among these policies, the carbon cap-and-trade system, as one of the most effective carbon reduction measures [5,14], has received extensive attention from scholars. Previous studies have indicated that changes in carbon trading prices and carbon quotas can have significant impacts on emission reduction [2,6]. As research has progressed further, scholars have started integrating the carbon cap-and-trade system with supply chain management. For instance, against the backdrop of the carbon cap-and-trade system, scholars have investigated inventory management [15], financing [15–17], joint replenishment [4,18], dual-channel sales [19], supply disruptions [20], product configuration [21], green innovation [22], competition [4,17,23], supply chain coordination [24], and supply chain design [25], among other issues. Additionally, there have been studies comparing the effectiveness of different carbon reduction policies by comparing the carbon cap-and-trade system with carbon tax policies [5,26–29]. However, ref. [26] explored the influence of different carbon reduction policies on enterprise technology selection and capacity decisions. Ref. [5] analyzed the effects of carbon taxes and carbon quota and trading policies on enterprise profits and emission reduction. Ref. [27] compared the effects of the carbon cap-and-trade system and carbon tax policies on multiproduct production planning for enterprises. Ref. [28] compared the effects of the carbon cap-and-trade system and carbon tax policies on trade credit. Ref. [29] compared the impact of two policies on emissions reductions from electric commercial vehicles (ECVs).

In the actual operation of supply chains, competition between supply chains can also threaten supply chain security. For instance, refs. [30,31] pointed out that manufacturers may be concerned about remanufacturing eroding their sales, leading to significant profit loss. Therefore, to address the security issues brought by competition, scholars have conducted research on competition between retailers [32,33], manufacturers [11,34], manufacturers and retailers [32,35] and supply chains [36]. Particularly concerning horizontal competition between manufacturers, researchers have employed various research methods to address such issues, such as the Cournot model [31,35], Nash equilibrium [37,38], and newsvendor model [11,39,40], among others. The difference lies in that the newsvendor model further considers the uncertainty of market demand and takes into account the substitutability of products between competing enterprises. When market demand is uncertain, although the above literature similarly uses the newsvendor model, ref. [39] studied the impact of sales commissions between two retailers on the optimal order quantity and profit. On the other hand, ref. [11] considered the effects of co-opetitive between two manufacturers under different substitution relationships.

As carbon emission reduction becomes increasingly emphasized, refs. [14,37,41-47]have further considered the issue of supply chain competition in a low-carbon environment. The difference among these studies lies in that ref. [41] did not specifically introduce carbon emission reduction policies while addressing the issue of interruptions combined with competition. Although refs. [37,46,47] all discuss financing issues, ref. [37] explores the impact of future carbon reduction policy stringency on green investment by competing firms, while refs. [46,47] only examine optimal financing strategies. Similar to ref. [41], refs. [37,46,47] also did not introduce specific carbon emission reduction policies. On the other hand, ref. [14] introduced carbon quota and trading policies and compared the profits and carbon emissions of supply chains under purely competitive and co-operative modes. Anand and ref. [42] compared the impacts of Carbon cap-and-trade system and carbon taxes on competition. Ref. [43] considered competition between multinational manufacturers and local manufacturers under carbon lock-in conditions. However, the aforementioned literature, due to its use of the Cournot model or Nash equilibrium, overlooks the uncertainty of market demand or the impact of product substitution on supply chain competition. In contrast to these studies, ref. [45] explores the factors influencing firms' green competitive advantage.

Unlike existing research, this paper considers the newsvendor competition problem under various substitution scenarios within a carbon cap-and-trade system. First, the paper explores the optimal decisions of two manufacturers under both purely competitive and co-opetitive modes. Then, we characterize the competition between manufacturers through different product substitution relationships and examine the impact of these relationships on the competing manufacturers. Third, we address the competition issue under market demand uncertainty by constructing a newsvendor model to analyze the problem. Finally, we incorporate the carbon cap-and-trade system and analyze the impact of low-carbon constraints on different scenarios to explore how the supply chain can achieve better benefits under various substitution situations.

3. Model Description

We will consider the competition between an original equipment manufacturer (OEM) and a contract manufacturer (CM) under the carbon cap-and-trade system. In this scenario, the final products manufactured by both the OEM and CM are sold in the same market, leading to a competitive relationship between them. Additionally, both the OEM and CM's final products generate carbon emissions, thereby subjecting both manufactures to carbon emission constraints. Furthermore, the CM has the capability to produce the raw materials required for the final products. Therefore, in this study, we will compare and analyze two competition modes as follows:

(1) Purely competitive mode: In this mode, the OEM procures raw materials from a third-party supplier (TS) for production and competes with the CM in the market for final products.

(2) Co-opetitive mode: In this mode, the OEM procures raw materials from the CM for production and competes with the CM in the market for final products.

At the beginning of the period, the enterprise acting as the supplier announces the wholesale prices of raw material w. Subsequently, OEM and CM, respectively, undertake their green investment k_o and k_c , and determine their yields q_o and q_c . Simultaneously, OEM orders raw materials from its supplier and pays wq_o for the supplier. At the end of the period, OEM and CM sell their respective final products at prices p and settle their carbon allowances through the carbon cap-and-trade system. If customers are unable to purchase products from OEM (or CM), they will attempt to purchase them from CM (or OEM). The parameters of this study are presented in Table 1.

Parameter	Definition	Parameter	Definition
С	unit production cost of raw materials	D_i	the random demand of manufacturer $i, i = o, c$
р	unit selling price of the final product	p_e	carbon trading price
<i>qi</i>	the yields of manufacturer i 's final product, $i = o, c$	δ_i	the demand growth caused by the green investment of manufacturer $i, i = o, c$
w_j	wholesale price from supplier $j, j = s, c$	ε_i	the demand uncertainty of manufacturer $i, i = o, c$
r _i	the proportion of manufacturer j's remaining demand allocated to manufacturer $i, i \neq j, i, j = o, c$	R_i	manufacturer <i>i</i> 's actual demand after substitution occurs, $i = o, c$
k_i	green investment of manufacturer $i, i = o, c$	e _i	the initial emission level of manufacturer <i>i</i> 's final product, $i = o, c$
E_i	carbon allowances of manufacturer $i, i = o, c$	$ heta_i$	the emission reduction rate per unit resulting from manufacturer i 's green investment, $i = o, c$
Subscript			
S	third-party supplier (TS)	С	contract manufacturer (CM)
0	original equipment manufacturer (OEM)		
Superscript			
Р	purely competitive mode	С	co-opetitive mode

Table 1. Parameters and definitions.

We assume that the random demand of manufacturer *i* is D_i (*i* = *o*, *c*) and is influenced by its green investment. Therefore, the demand function is $D_i = \delta_i \sqrt{k_i} + \varepsilon_i$ [48], in which δ_i represents the increase in demand per unit of green investment, and ε_i is a non-negative random variable. Its distribution function and density function on $[0, +\infty)$ are $F(\varepsilon_i)$ and $f(\varepsilon_i)$, respectively. Let $\overline{F}(\varepsilon_i) = 1 - F(\varepsilon_i)$ be the reliability function of ε_i . The increasing generalized failure rate of ε_i is $g(\varepsilon_i) = \varepsilon_i h(\varepsilon_i)$ [49], where $h(\varepsilon_i) = f(\varepsilon_i)/\bar{F}(\varepsilon_i)$ is the failure rate. Both $h(\varepsilon_i)$ and $g(\varepsilon_i)$ are monotonically increasing with ε_i [50]. We assume that D_o and D_c are mutually independent and follow the same distribution [11]. We also adhere to the product substitution assumptions outlined in [40], which relate the degree of product substitution to the level of customer purchase loyalty. When customer loyalty to a manufacturer is low, its products are more easily substituted. We assume that when one party cannot meet all the demand, a certain proportion of the remaining demand will be transferred to the competitor. When purchase loyalty is zero, one-way substitution will occur [13]. Therefore, when CM cannot satisfy its demand, the actual demand R_o for OEM is $R_o = D_o + r_o (D_c - q_c)^+$, where r_o is the proportion of CM's remaining demand allocated to OEM. Similarly, when OEM cannot satisfy its demand, the actual demand R_c for CM is r_c is the proportion of OEM's remaining demand allocated to CM. Additionally, we assume that the production cost of raw materials is c. The procurement cost of non-critical components and the production cost of final products are zero [11], where $c + p_e(e_c - \theta_c k_c) < p$. Furthermore, to simplify the problem, we assume that no carbon emissions occur during the production process of raw materials. The wholesale price *w* satisfies $c < w \leq p - p_e(e_o - \theta_o k_o)$.

4. Model Analysis of Purely Competitive Supply Chain

In this section, we first consider the scenario where OEM and CM are in a state of pure competition. The supply chain structure of the purely competitive mode is illustrated in Figure 1.



Figure 1. Supply chain structure of the purely competitive mode.

At the beginning of the period, TS announces the wholesale price of raw materials w_s . Then, OEM and CM undertake their respective green investments k_o and k_c , and determine their yields q_o and q_c . Meanwhile, OEM orders raw materials from TS and pays TS $w_s q_o$. After undertaking green investments, the unit carbon emissions of the final products of OEM and CM, respectively, decrease to $(e_o - \theta_o k_o)$ and $(e_c - \theta_c k_c)$. At the end of the period, the final products of OEM and CM are sold at prices p, and they, respectively, obtain carbon trading revenues of $p_e[E_o - (e_o - \theta_o k_o)q_o]$ and $p_e[E_c - (e_c - \theta_c k_c)q_c]$ through the carbon cap-and-trade system. The sequence of events is illustrated in Figure 2.



Figure 2. Sequence of events under purely competitive mode.

On this basis, we will analyze the purely competitive mode under different product substitution scenarios.

4.1. Purely Competitive Mode: Non-Substitutable (0, 0)

When there is no substitutability between the final products of OEM and CM, it satisfies $r_o = r_c = 0$. Therefore, the actual demand for OEM and CM equals their respective random demands, i.e., $R_o = D_o$, $R_c = D_c$. In this case, the expected sales revenue for OEM and CM at the end of the period are $pE[\min[D_o, q_o]]$ and $pE[\min[D_c, q_c]]$, respectively.

The profit functions for TS, OEM and CM in this scenario are as follows:

$$\pi_s^P = (w_s - c)q_o \tag{1}$$

$$\pi_o^P = pE[\min[D_o, q_o]] - w_s q_o + p_e[E_o - (e_o - \theta_o k_o)q_o] - k_o$$
⁽²⁾

$$\pi_{c}^{P} = pE[\min[D_{c}, q_{c}]] - cq_{c} + p_{e}[E_{c} - (e_{c} - \theta_{c}k_{c})q_{c}] - k_{c}$$
(3)

Lemma 1. Under the (0, 0) scenario in the purely competitive mode, the optimal decisions for TS, OEM and CM are satisfied by: (All proofs in Supplementary Materials)

$$w_s^{P(0,0)} = pq_o^{P(0,0)} f(q_o^{P(0,0)} - \delta_o \sqrt{k_o}) + c$$
$$q_o^{P(0,0)} = F^{-1}(\frac{p - w_s^{P(0,0)} - p_e(e_o - \theta_o k_o)}{p}) + \delta_o \sqrt{k_o}$$
$$q_c^{P(0,0)} = F^{-1}(\frac{p - c - p_e(e_c - \theta_c k_c)}{p}) + \delta_c \sqrt{k_c}$$

From Lemma 1, we observe that under purely competitive mode, when there is no substitution between the products of OEM and CM, the optimal order quantities for both are independent of each other. Building upon Lemma 1, in Proposition 1, we provide the impact of carbon trading price p_e and the green investments k_o and k_c on the optimal decisions.

Proposition 1. Under the (0, 0) scenario in the purely competitive mode, (1) $q_o^{P(0,0)}$ and $w_o^{P(0,0)}$ increase with the increase in k_o , while $q_c^{P(0,0)}$ remains unchanged with variations in k_o ; (2) $q_c^{P(0,0)}$ increases with the increase in k_c , while $q_o^{P(0,0)}$ and $w_o^{P(0,0)}$ remains unchanged with variations in k_c ; (3) $q_c^{P(0,0)}$, $q_o^{P(0,0)}$ and $w_o^{P(0,0)}$ decrease with the increase in p_e .

In Proposition 1, we observe that $q_o^{P(0,0)}$ and $w_o^{P(0,0)}$ are unaffected by CM's green investment, while $q_c^{P(0,0)}$ is unaffected by OEM's green investment. This is because there is no substitution relationship between the final products of OEM and CM. The growth in green investment in one manufacturer does not affect the demand of the other manufacturer. Additionally, the manufacturers' green investment can increase the demand; hence, the optimal production quantities of OEM and CM increase with their respective green investments. For TS, an increase in OEM's green investment leads to higher demand, prompting TS to raise wholesale prices to enhance its revenue. However, the impact of carbon trading prices on optimal decisions differs. With an increase in carbon trading prices, OEM and CM reduce their production quantities to maximize carbon trading revenue. As OEM's production decreases, TS lowers wholesale prices to stimulate OEM to increase its order quantity and ensure its revenue.

To simplify subsequent analyses, we assume that ε_i (i = o, c) follows a uniform distribution from 0 to 1 [51]. Under this assumption, we employ backward induction, and in Lemma 2, we provide the conditions for the optimal solutions of TS, OEM and CM.

Lemma 2. Under the (0, 0) scenario in the purely competitive mode, when ε_i (i = o, c) follows a uniform distribution from 0 to 1, the optimal decisions for TS, OEM and CM are satisfied by:

$$w_{s}^{P(0,0)} = \frac{p(1+\delta_{o}\sqrt{k_{o}}) + c - p_{e}(e_{o} - \theta_{o}k_{o})}{2}$$
$$q_{o}^{P(0,0)} = \frac{p(1+\delta_{o}\sqrt{k_{o}}) - c - p_{e}(e_{o} - \theta_{o}k_{o})}{2p}$$
$$q_{c}^{P(0,0)} = 1 + \delta_{c}\sqrt{k_{c}} - \frac{c + p_{e}(e_{c} - \theta_{c}k_{c})}{p}$$

4.2. Purely Competitive Mode: OEM Substitutable for CM (r, 0)

When OEM products partially substitute for CM products, we have $r_o = r \neq 0$ and $r_c = 0$. In this case, the actual demand for OEM is $R_o = D_o + r(D_c - q_c)^+$, while the actual demand for CM is equal to its random demand, i.e., $R_c = D_c$. Consequently, the expected sales revenue for OEM and CM at the end of the period are $pE[\min[D_o + r(D_c - q_c)^+, q_o]]$

$$\pi_s^P = (w_s - c)q_o \tag{4}$$

$$\pi_o^P = pE[\min[D_o + r(D_c - q_c)^+, q_o] - w_s q_o + p_e[E_o - (e_o - \theta_o k_o)q_o] - k_o$$
(5)

$$\pi_{c}^{P} = pE[\min[D_{c}, q_{c}]] - cq_{c} + p_{e}[E_{c} - (e_{c} - \theta_{c}k_{c})q_{c}] - k_{c}$$
(6)

Next, we denote $L = \frac{\partial \pi_s}{\partial w_s}$ and provide the conditions satisfied by the optimal solutions of TS, OEM and CM in Lemma 3.

Lemma 3. In the case of (r, 0) under purely competitive mode,

(1) Under any conditions, $q_c^{P(r,0)} = 1 + \delta_c \sqrt{k_c} - \frac{c + p_e(e_c - \theta_c k_c)}{p}$; (2) When $e_o - \theta_o k_o < M_1$,

$$q_o^{P(r,0)} = \frac{1 + \delta_o \sqrt{k_o}}{2} + \frac{r[c + p_e(e_c - \theta_c k_c)]^2}{4p^2} - \frac{c}{2p} - \frac{p_e(e_o - \theta_o k_o)}{2p}$$
$$w_s^{P(r,0)} = \frac{p(1 + \delta_o \sqrt{k_o})}{2} + \frac{r[c + p_e(e_c - \theta_c k_c)]^2}{4p} + \frac{c}{2} - \frac{p_e(e_o - \theta_o k_o)}{2}$$

(3) When $M_1 \le e_o - \theta_o k_o < M_2$, $w_s^{P(r,0)}$ satisfies $L(w_s^{P(r,0)}) = 0$, and $q_o^{P(r,0)}$ satisfies

$$q_o^{P(r,0)} = \sqrt{r^2 \left[1 - \frac{c + p_e(e_c - \theta_c k_c)}{p}\right]^2 + 2r \left[1 - \frac{w_s^{P(r,0)} + p_e(e_o - \theta_o k_o)}{p}\right] + \delta_o \sqrt{k_o} - r \left[1 - \frac{c + p_e(e_c - \theta_c k_c)}{p}\right]}$$

(4) When
$$e_o - \theta_o k_o \ge M_2$$
, $q_o^{P(r,0)} = \delta_o \sqrt{k_o}$, $w_s^{P(r,0)} = p - p_e(e_o - \theta_o k_o)$.
where $M_1 = \frac{p}{p_e} \{ \frac{r[c+p_e(e_c - \theta_c k_c)]^2}{2p^2} - \frac{c}{p} - \frac{2r[c+p_e(e_c - \theta_c k_c)]}{p} + 1 - \delta_o \sqrt{k_o} \}$, $M_2 = \frac{1}{p_e} [(p-c)(1 - \delta_o \sqrt{k_o}) + \delta_o \sqrt{k_o} p_e(e_c - \theta_c k_c)]$.

Next, we first compare the optimal solutions for the (r, 0) and (0, 0) scenarios in the purely competitive mode in Proposition 2.

Proposition 2. Comparing the (r, 0) and (0, 0) scenarios in the purely competitive mode, we obtain: (1) Under any conditions, $q_c^{P(r,0)} = q_c^{P(0,0)}$, $\pi_c^{P(r,0)} = \pi_c^{P(0,0)}$;

(2) When
$$e_o - \theta_o k_o < M_1$$
, $q_o^{P(r,0)} > q_o^{P(0,0)}$, $w_s^{P(r,0)} > w_s^{P(0,0)}$, $\pi_s^{P(r,0)} = \pi_s^{P(0,0)}$;
(3) When $e_o - \theta_o k_o \ge M_2$, $q_o^{P(r,0)} < q_o^{P(0,0)}$, $w_s^{P(r,0)} > w_s^{P(0,0)}$.

From Proposition 2, we observe that in the (r, 0) scenario, the optimal production quantity and profit of CM are the same as in the (0, 0) scenario. This is because CM's products cannot substitute for OEM's products, leading to no additional demand. Therefore, CM's decision-making process does not need to consider the influence of OEM. For OEM and TS, the optimal decisions in this scenario are influenced by the unit carbon emissions of OEM's products. When the carbon emissions of OEM's products are low $(e_o - \theta_o k_o < M_1)$, compared to $q_o^{P(0,0)}$ and $w_s^{P(0,0)}$, both the production quantity of OEM and the wholesale price of TS will increase. This is because some demand that CM cannot fulfill will shift to OEM's market. Additionally, the increase in sales revenue due to increased production compensates for the loss of carbon trading revenue. In this case, raising the wholesale price by TS will not significantly reduce the ordering quantity of OEM's products are sufficiently high $(e_o - \theta_o k_o \ge M_2)$, OEM will produce at the minimum quantity to maximize profit. Similarly, TS will set the wholesale price at the maximum to ensure its profit.

Proposition 3. *In the* (*r*, 0) *scenario under purely competitive mode,*

(1) Under any conditions, $q_{L}^{P(r,0)}$ remains constant regardless of the change in r;

2) When
$$e_0 - \theta_0 k_0 < M_1$$
, $q_0^{P(r,0)}$ and $w_s^{P(r,0)}$ increase with the increase in r;

(3) When $M_1 \leq e_o - \theta_o k_o < M_2$, $q_o^{P(r,0)}$ increases with the increase in r; if $rq_o^{P(r,0)}(1 - \frac{c+p_e(e_c - \theta_c k_c)}{p}) > [r(1 - \frac{c+p_e(e_c - \theta_c k_c)}{p}) + q_o^{P(r,0)} - \delta_o \sqrt{k_o}]^2$, $w_s^{P(r,0)}$ increases with the increase in r;

(4) When $e_o - \theta_o k_o \ge M_2$, $q_o^{P(r,0)}$ and $w_s^{P(r,0)}$ remain constant regardless of the change in r.

From Proposition 3, we observe that regardless of how the substitution rate r changes, the optimal production quantity for CM remains constant. This is because in this scenario, CM's actual demand equals its stochastic demand, and CM cannot substitute for the excess demand of OEM. Therefore, when making decisions, CM does not need to consider the influence of the substitution rate r. However, for OEM and TS, when the carbon emission per unit of OEM's product is sufficiently low ($e_0 - \theta_0 k_0 < M_1$), the cost per unit for OEM is low, and increasing its production can lead to greater profits. Since the cost per unit for OEM is low, raising the wholesale price by TS not only does not result in a significant reduction in OEM's production but rather brings more profits to TS. But when the carbon emission per unit of OEM's product is sufficiently high $(e_0 - \theta_0 k_0 \ge M_2)$, the cost per unit for OEM is correspondingly high. To ensure its profit, OEM will always produce at the minimum quantity. For TS to ensure its profit, it will always set the wholesale price at the maximum. When the carbon emission per unit of OEM's product is at an intermediate level ($M_1 \leq e_0 - \theta_0 k_0 < M_2$), there is still some profit margin for OEM, and increasing its production can yield greater returns. However, in this scenario, the wholesale price set by TS no longer monotonically changes with the substitution rate, as it is influenced by the carbon emission per unit of OEM and CM's products, as well as the carbon trading price.

Proposition 4. In the (r, 0) scenario under purely competitive mode,

(1) Under any conditions, $q_c^{P(r,0)}$ remains constant regardless of the change in k_c and decreases with the increase in p_e , and remains unchanged with the variation in k_o ;

(2) When $e_o - \theta_o k_o < M_1$, $q_o^{P(r,0)}$ and $w_s^{P(r,0)}$ increase with the increase in k_o and decreases with the increase in k_c ; if $p_e > \frac{1}{e_c - \theta_c k_c} \left[\frac{p(e_o - \theta_o k_o)}{r(e_c - \theta_c k_c)} - c \right]$, $q_o^{P(r,0)}$ and $w_s^{P(r,0)}$ increase with the increase in p_e ;

(3) When $M_1 \leq e_o - \theta_o k_o < M_2$, $q_o^{P(r,0)}$ and $w_s^{P(r,0)}$ increase with the increase in k_o and decreases with the increase in k_c ; if $\frac{e_o - \theta_o k_o}{(2q_o^{P(r,0)} - \delta_o \sqrt{k_o})(e_c - \theta_c k_c)} > \frac{q_o^{P(r,0)} - \delta_o \sqrt{k_o} + r[1 - \frac{c + pe(e_c - \theta_c k_c)}{p}]}{2q_o^{P(r,0)} - \delta_o \sqrt{k_o} + r[1 - \frac{c + pe(e_c - \theta_c k_c)}{p}]}$, $q_o^{P(r,0)}$ increase with the increase in k_c ; if $k_c > \frac{p(q_o^{P(r,0)} - \delta_o \sqrt{k_o})^2}{\delta_o \sqrt{k_o r p_e \theta_c}} - \frac{p - c - p_e e_c}{p_e \theta_c}$, $w_s^{P(r,0)}$ increase with the increase in k_c ; if $\frac{[1 - \frac{c + pe(e_c - \theta_c k_c)}{p}] \{r[1 - \frac{c + pe(e_c - \theta_c k_c)}{p}] - \delta_o \sqrt{k_o}\}}{q_o^{P(r,0)} - \delta_o \sqrt{k_o} + r[1 - \frac{c + pe(e_c - \theta_c k_c)}{p}] - \delta_o \sqrt{k_o}\}} > \frac{2}{r} [\frac{e_o - \theta_o k_o}{e_c - \theta_c k_c} - (q_o^{P(r,0)} - \delta_o \sqrt{k_o})]$, $w_s^{P(r,0)}$ increase with the increase in p_e ;

(4) When $e_o - \theta_o k_o \ge M_2$, $q_o^{P(r,0)}$ increases with the increase in k_o and remains unchanged with changes in k_c and p_e ; $w_s^{P(r,0)}$ increase with the increase in k_o , decreases with the increase in p_e , and remains unchanged with changes in k_c .

From Proposition 4, we can see that the impact of carbon trading price p_e and the green investment of both manufactures, k_o and k_c , on the optimal yields of CM is the same as in the case of a purely competitive mode (0, 0). This is because in this scenario, the actual demand of CM is equal to its stochastic demand. Additionally, the effect of OEM's green investment k_o on $q_o^{P(r,0)}$ and $w_s^{P(r,0)}$ is also similar to the purely competitive scenario (0, 0). This is because OEM's green investment k_o brings an increase in demand. Furthermore, an increase in OEM's green investment brings more demand to TS. To increase its revenue, TS raises the wholesale price. When OEM's carbon emissions are low enough ($e_o - \theta_0 k_o < M_1$),

increasing CM's green investment k_c can increase its yields, reducing the excess demand that cannot be satisfied. Consequently, OEM's optimal yields decrease. Additionally, because OEM's carbon emissions are low in this scenario, when carbon trading prices are low, OEM's carbon trading revenue will be severely affected. To obtain more carbon trading revenue, OEM will also reduce its yields. Due to the decrease in OEM's yields, TS will lower the wholesale price to ensure its profit. When OEM's carbon emissions are correspondingly high ($e_o - \theta_o k_o \ge M_2$), the cost per unit of OEM's product is high. Therefore, to ensure its profit, OEM produces at the minimum yields, and TS charges the highest wholesale price. When OEM's carbon emissions satisfy $M_1 \le e_o - \theta_o k_o < M_2$, the impact of carbon trading price p_e on $q_o^{P(r,0)}$ and $w_s^{P(r,0)}$ is enough complex due to the influence of other parameters. This phenomenon indicates that for the manufacturer constrained by emission reduction, an increase in carbon trading price does not necessarily suppress his production capacity.

4.3. Purely Competitive Mode: CM Substitutable for OEM (0, r)

When CM products partially substitute for OEM products, it satisfies $r_c = r \neq 0$ and $r_o = 0$. In this case, the actual demand for OEM equals its random demand, i.e., $R_o = D_o$. Meanwhile, CM's actual demand is represented by $R_c = D_c + r(D_o - q_o)^+$. Consequently, the expected sales revenue for OEM and CM at the end of the period are $pE[\min[D_o, q_o]]$ and $pE[\min[D_c + r(D_o - q_o)^+, q_c]$, respectively. The profit functions for TS, OEM and CM under this scenario are as follows:

$$\pi_s^P = (w_s - c)q_o \tag{7}$$

$$\pi_o^P = pE[\min[D_o, q_o]] - w_s q_o + p_e[E_o - (e_o - \theta_o k_o)q_o] - k_o$$
(8)

$$\pi_c^P = pE[\min[D_c + r(D_o - q_o)^+, q_c] - cq_c + p_e[E_c - (e_c - \theta_c k_c)q_c] - k_c$$
(9)

Lemma 4. *In the case of (0, r) under purely competitive mode,*

(1) Under any conditions,

$$q_o^{P(0,r)} = \frac{p(1+\delta_o\sqrt{k_o}) - c - p_e(e_o - \theta_o k_o)}{2p}$$
$$w_s^{P(0,r)} = \frac{p(1+\delta_o\sqrt{k_o}) + c - p_e(e_o - \theta_o k_o)}{2}$$

(2) When $e_c - \theta_c k_c \leq H_1$,

$$q_{c}^{P(0,r)} = \frac{r[p(1+\delta_{o}\sqrt{k_{o}})+c+p_{e}(e_{o}-\theta_{o}k_{o})]}{2p} + 1 + \delta_{c}\sqrt{k_{c}} - \sqrt{\frac{2r[c+p_{e}(e_{c}-\theta_{c}k_{c})]}{p}}$$

(3) When
$$H_1 < e_c - \theta_c k_c < H_2$$

$$q_c^{P(0,r)} = 1 + \delta_c \sqrt{k_c} - \frac{c + p_e(e_c - \theta_c k_c)}{p} + \frac{r}{2} \left[\frac{p(1 + \delta_o \sqrt{k_o}) + c + p_e(e_o - \theta_o k_o)}{2p}\right]^2$$

(4) When
$$e_c - \theta_c k_c \ge H_2$$

$$q_{c}^{P(0,r)} = \sqrt{r^{2} \left[\frac{p(1-\delta_{o}\sqrt{k_{o}}) - c - p_{e}(e_{o} - \theta_{o}k_{o})}{2p}\right]^{2} + 2r\left[1 - \frac{c + p_{e}(e_{c} - \theta_{c}k_{c})}{p}\right]} + \delta_{c}\sqrt{k_{c}} - \frac{r}{2p}\left[p(1-\delta_{o}\sqrt{k_{o}}) - c - p_{e}(e_{o} - \theta_{o}k_{o})\right]}$$

$$where H_{1} = \frac{r\left[p(1+\delta_{o}\sqrt{k_{o}}) + c + p_{e}(e_{o} - \theta_{o}k_{o})\right]^{2}}{8p_{e}p} - \frac{c}{p_{e}}, H_{2} = \frac{r}{2p_{e}}\left[p(1-\delta_{o}\sqrt{k_{o}}) - c - p_{e}(e_{o} - \theta_{o}k_{o})\right] + (1-r)\frac{p}{p_{e}} + \frac{r\left[p(1+\delta_{o}\sqrt{k_{o}}) + c + p_{e}(e_{o} - \theta_{o}k_{o})\right]^{2}}{8p_{e}p} - \frac{c}{p_{e}}.$$

Proposition 5. Comparing the optimal decisions under purely competitive for the (0, r) and (0, 0) scenarios, we obtain: (1) Under any conditions, $q_o^{P(0,r)} = q_o^{P(0,0)}$, $w_s^{P(0,r)} = w_s^{P(0,0)}$, $\pi_o^{P(0,r)} = \pi_s^{P(0,0)}$, $\pi_s^{P(r,0)} = \pi_s^{P(0,0)}$; (2) When $e_c - \theta_c k_c < H_2$, $q_c^{P(0,r)} > q_c^{P(0,0)}$; otherwise, $a_{c}^{\tilde{P}(0,r)} < a_{c}^{\tilde{P}(0,0)}$.

From Proposition 5, we observe that the optimal decisions and profits of OEM and TS in this scenario are the same as those in the purely competitive mode under the (0, 0)scenario. This is because OEM's products cannot substitute for CM's products, resulting in no additional demand. Therefore, OEM and TS do not need to consider the influence of CM when making decisions. However, CM's decision is no longer uniform. Similar to the decision-making process of OEM in the purely competitive mode under the (r, 0) scenario, CM needs to consider the carbon emissions of its products in this scenario. When the carbon emissions per unit of CM's product satisfy $e_c - \theta_c k_c < H_2$, CM's order quantity will accept some of OEM's excess demand. In this case, CM's optimal production quantity will exceed the optimal production quantity under the (0, 0) scenario in the purely competitive mode. However, when the carbon emissions per unit of CM's product satisfy $e_c - \theta_c k_c \ge H_2$, the cost of CM's unit product becomes sufficiently high. In this case, CM will no longer accept some of OEM's excess demand. To ensure its profit, CM's production quantity will be smaller than the optimal production quantity under the (0, 0) scenario in the purely competitive mode.

Proposition 6. In the purely competitive mode under the (r, 0) scenario, $\frac{\partial q_0^{P(0,r)}}{\partial r} = 0$, $\frac{\partial w_s^{P(0,r)}}{\partial r} = 0$, $\frac{\partial q_c^{P(\bar{0},r)}}{\partial r} > 0.$

Proposition 6 illustrates that under any conditions, $q_o^{P(0,r)}$ and $w_s^{P(0,r)}$ are not influenced by the substitution rate r. Since OEM's actual demand equals its random demand, and only part of the excess demand from OEM is substituted by CM, OEM and TS do not need to consider this factor in their decision-making. As for CM, according to Proposition 5, when CM's product carbon emissions are sufficiently high $(e_c - \theta_c k_c \ge H_2)$, its optimal yields $q_c^{P(0,r)}$ will be less than $q_c^{P(0,0)}$. Therefore, as CM's unit product carbon emissions increase, its optimal yields may decrease. For instance, when $e_c - \theta_c k_c \ge H_2$, CM's optimal yields are less than in the other two stages. However, within each stage, its optimal yields increase with the substitution rate r. This is because CM produces its raw materials and does not need to order from another supplier, and its unit production cost remains constant. Thus, regardless of how much CM's unit product carbon emissions increase, there will always be a certain profit margin per unit product. Additionally, an increase in the substitution rate r can bring more demand to CM. Therefore, CM's optimal yields increase with the increase in the substitution rate r. This phenomenon illustrates that when a manufacturer can independently produce raw materials, the higher the substitution rate of its products with those of competitors, the higher its optimal yields will be.

Proposition 7. In the purely competitive mode under the (r, 0) scenario, (1) Under any conditions, $\frac{\partial q_o^{P(0,r)}}{\partial k_o} > 0$, $\frac{\partial w_s^{P(0,r)}}{\partial k_o} > 0$, $\frac{\partial q_o^{P(0,r)}}{\partial p_e} < 0$, $\frac{\partial w_s^{P(0,r)}}{\partial p_e} < 0$, $\frac{\partial q_c^{P(0,r)}}{\partial k_c} > 0$; if $k_o < (\frac{p\delta_o}{2p_e\theta_o})^2$, $\frac{\partial q_c^{P(0,r)}}{\partial k_o} > 0$; (2) When $e_c - \theta_c k_c \leq H_1$, if $p_e > \frac{2p(e_c - \theta_c k_c)}{r} - \frac{c}{e_c - \theta_c k_c}$, $\frac{\partial q_c^{P(0,r)}}{\partial p_e} > 0$; (3) When $H_1 < e_c - \theta_c k_c < H_2$, if $p_e > \frac{1}{e_o - \theta_o k_o} [\frac{2(e_c - \theta_c k_c)}{r(e_o - \theta_o k_o)} - p(1 + \delta_o \sqrt{k_o}) - c]$, $\frac{\partial q_c^{P(0,r)}}{\partial p_e} > 0$; (4) When $e_c - \theta_c k_c \ge H_2$, if $p_e[(e_o - \theta_o k_o) - (e_c - \theta_c k_c)] > \frac{2p(e_c - \theta_c k_c)^2}{r(e_o - \theta_o k_o)^2} - p\delta_o \sqrt{k_o}$, $\frac{\partial q_c^{P(0,r)}}{\partial p_e} > 0.$

Proposition 7 indicates that $q_o^{P(0,r)}$ and $w_o^{P(0,0)}$ are unaffected by CM's green investment. This is because OEM's products cannot substitute for CM's products. However, manufacturer's green investments can lead to an increase in demand, thus causing optimal production quantities for both OEM and CM to increase with their respective green investments. Regarding TS, an increase in OEM's green investment results in greater demand. To enhance revenue, TS raises wholesale prices. As carbon trading prices increase, OEM and CM reduce their production to maximize carbon trading revenue. With a decrease in OEM's production, TS lowers wholesale prices to stimulate increased ordering by OEM and maintain revenue. In addition, for CM, regardless of its carbon emissions stage, the optimal yields of CM increase with an increase in k_o only when OEM's green investment is relatively low. This is because when OEM's green investment is low, it fails to satisfy the excess demand. CM will increase its yields to satisfy a greater substitutable demand. Furthermore, when the carbon emissions per unit product of CM are low ($e_c - \theta_c k_c < H_2$), a low carbon trading price will harm CM's carbon trading revenue. CM will reduce production to obtain more carbon trading revenue. When the carbon emissions per unit product of CM are sufficiently high $(e_c - \theta_c k_c \ge H_2)$, the impact of carbon trading price on CM's yields need to be evaluated in conjunction with the carbon emissions per unit product of both CM and OEM.

4.4. Purely Competitive Mode: CM and OEM Symmetrically Substitutable (r, r)

When CM and OEM products can be symmetrically substituted, we have $r_c = r_o = r \neq 0$. In this scenario, the actual demands for OEM and CM are, respectively, $R_o = D_o + r(D_c - q_c)^+$ and $R_c = D_c + r(D_o - q_o)^+$. In the end, the expected sales revenue for OEM and CM are $pE[\min[D_o + r(D_c - q_c)^+, q_o]]$ and $pE[\min[D_c + r(D_o - q_o)^+, q_c]]$, respectively. In this case, the profit functions for TS, OEM, and CM are as follows:

$$\pi_s^P = (w_s - c)q_o \tag{10}$$

$$\pi_o^P = pE[\min[D_o + r(D_c - q_c)^+, q_o] - w_s q_o + p_e[E_o - (e_o - \theta_o k_o)q_o] - k_o$$
(11)

$$\pi_c^P = pE[\min[D_c + r(D_o - q_o)^+, q_c] - cq_c + p_e[E_c - (e_c - \theta_c k_c)q_c] - k_c$$
(12)

Due to the complexity of solving this scenario, to simplify the problem, we will consider the special case of r = 1, where there is complete product substitution between OEM and CM. In this case, market demand that OEM (or CM) cannot meet will be completely transferred to CM (or OEM). This situation also exists in reality. We provide the conditions for the optimal solutions of TS, OEM, and CM in Lemma 5.

Lemma 5. In the scenario of a purely competitive mode under (1, 1) conditions,

(1) When $\frac{w_s + p_e(e_o - \theta_o k_o)}{p} \leq \frac{1}{2} [\frac{c + p_e(e_c - \theta_c k_c)}{p}]^2$, $q_c^{P(1,1)} = 1 + \delta_c \sqrt{k_c} - \frac{c + p_e(e_c - \theta_c k_c)}{p}$, the optimal wholesale price satisfies $L(w_s^{P(1,1)}) = 0$, and optimal yield for OEM satisfies:

$$q_o^{P(1,1)} = 1 + \delta_o \sqrt{k_o} + \frac{c + p_e(e_c - \theta_c k_c)}{p} - \sqrt{\frac{2[w_s^{P(1,1)} + p_e(e_o - \theta_o k_o)]}{p}}$$

(2) When $\frac{1}{2} \left[\frac{c+p_e(e_c-\theta_c k_c)}{p}\right]^2 < \frac{w_s+p_e(e_o-\theta_o k_o)}{p} < \sqrt{\frac{2[c+p_e(e_c-\theta_c k_c)]}{p}}$, if $q_c - \delta_c \sqrt{k_c} + \frac{q_o-\delta_o \sqrt{k_o}}{r} > 1$, the optimal wholesale price satisfies $L(w_s^{P(1,1)}) = 0$, and the optimal yields, respectively, satisfy:

$$8[1 + \delta_c \sqrt{k_c} - q_c^{P(1,1)} - \frac{c + p_e(e_c - \theta_c k_c)}{p}] = -\left\{\frac{2[w_s^{P(1,1)} + p_e(e_o - \theta_o k_o)]}{p} - \left[1 - (q_c^{P(1,1)} - \delta_c \sqrt{k_c})\right]^2\right\}^2$$

8

$$\begin{split} [1+\delta_{o}\sqrt{k_{o}}-q_{o}^{p(1,1)}-\frac{w_{s}^{P(1,1)}+p_{c}(e_{o}-\theta_{o}k_{o})}{p}] &= -\{\frac{2[c+p_{e}(e_{c}-\theta_{c}k_{c})]}{p}-[1-(q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}})]^{2}\}^{2} \\ & (3) \quad When \quad \frac{1}{2}[\frac{c+p_{c}(e_{c}-\theta_{c}k_{c})}{p}]^{2} \quad < \frac{w_{s}+p_{e}(e_{o}-\theta_{o}k_{o})}{p} \quad < \sqrt{\frac{2[c+p_{e}(e_{c}-\theta_{c}k_{c})]}{p}}, \\ & if q_{c}-\delta_{c}\sqrt{k_{c}}+\frac{q_{o}-\delta_{o}\sqrt{k_{o}}}{r} \leq 1, the optimal wholesale price satisfies $L(w_{s}^{P(1,1)}) = 0, and the optimal yields, respectively, satisfy: \\ & \frac{q_{c}^{P(1,1)}-\delta_{c}\sqrt{k_{c}}}{2} + \frac{p-c-p_{e}(e_{c}-\theta_{c}k_{c})}{p(q_{c}^{P(1,1)}-\delta_{c}\sqrt{k_{c}})} = \sqrt{(q_{c}^{P(1,1)}-\delta_{c}\sqrt{k_{c}})^{2}+2[1-\frac{w_{s}^{P(1,1)}+p_{e}(e_{o}-\theta_{o}k_{o})}{p}] \\ & \frac{q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}}}{2} + \frac{p-w_{s}^{P(1,1)}-p_{e}(e_{o}-\theta_{o}k_{o})}{p(q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}})} = \sqrt{(q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}})^{2}+2[1-\frac{c+p_{e}(e_{c}-\theta_{c}k_{c})}{p}] \\ & \frac{q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}}}{2} + \frac{p-w_{s}^{P(1,1)}-p_{e}(e_{o}-\theta_{o}k_{o})}{p(q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}})} = \sqrt{(q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}})^{2}+2[1-\frac{c+p_{e}(e_{c}-\theta_{c}k_{c})}{p}] \\ & \frac{q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}}}{2} + \frac{p-w_{s}^{P(1,1)}-p_{e}(e_{o}-\theta_{o}k_{o})}{p(q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}})} = \sqrt{(q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}})^{2}+2[1-\frac{c+p_{e}(e_{c}-\theta_{c}k_{c})}{p}] \\ & \frac{q_{o}^{P(1,1)}-2+\delta_{c}\sqrt{k_{c}}}{p(q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}})} = \sqrt{(q_{o}^{P(1,1)}-\delta_{o}\sqrt{k_{o}})^{2}+2[1-\frac{c+p_{e}(e_{c}-\theta_{c}k_{c})}{p}] \\ & \frac{q_{o}^{P(1,1)}=2+\delta_{c}\sqrt{k_{c}}-\sqrt{\frac{2[c+p_{e}(e_{c}-\theta_{c}k_{c})]}{p}}, if \delta_{o}\sqrt{k_{o}}+\frac{p_{e}(e_{o}-\theta_{o}k_{o})+c}{p}} \geq 1, \\ & q_{c}^{P(1,1)}=1+\delta_{c}\sqrt{k_{c}}-\sqrt{\frac{2[c+p_{e}(e_{c}-\theta_{c}k_{c})]}{p}}, if \delta_{o}\sqrt{k_{o}}+\frac{p_{e}(e_{o}-\theta_{o}k_{o})+c}{p}} < 1, \\ & q_{o}^{P(1,1)}=1+\delta_{c}\sqrt{k_{c}}-\sqrt{\frac{2[c+p_{e}(e_{c}-\theta_{c}k_{c})]}{p}} + \frac{1}{2p}[p(1+\delta_{o}\sqrt{k_{o}})+c+p_{e}(e_{o}-\theta_{o}k_{o})] \\ & w_{s}^{P(1,1)}=\frac{1}{2}[p(1+\delta_{o}\sqrt{k_{o}})+c-p_{e}(e_{o}-\theta_{o}k_{o})] \end{cases}$$$

Through Lemma 5, we find that when r = 1, the problem of (r, r) in purely competitive mode can be simplified into five cases. However, the optimal solution in the (1, 1) scenario ceases to be an implicit function only when $\frac{w_s + p_e(e_o - \theta_o k_o)}{p} \ge \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}}$. Therefore, building upon Lemma 5, we present in Proposition 8 a comparison of certain (1, 1) scenarios and the (0, 0) optimal solution under the purely competitive mode.

Proposition 8. Comparing the optimal decisions in the (1, 1) and (0, 0) scenarios under the purely competitive mode, we obtain:

 $\begin{array}{l} \text{(1) When } \frac{w_s + p_e(e_o - \theta_o k_o)}{p} \geq \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}} \text{ and } \delta_o \sqrt{k_o} + \frac{p_e(e_o - \theta_o k_o) + c}{p} \geq 1 \text{ are satisfied,} \\ q_c^{P(1,1)} > q_c^{P(0,0)}, q_o^{P(1,1)} \leq q_o^{P(0,0)}, w_s^{P(1,1)} \geq w_s^{P(0,0)}; \\ \text{(2) When } \frac{w_s + p_e(e_o - \theta_o k_o)}{p} \geq \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}} \text{ and } \delta_o \sqrt{k_o} + \frac{p_e(e_o - \theta_o k_o) + c}{p} < 1 \text{ are satisfied,} \\ q_o^{P(1,1)} = q_o^{P(0,0)}, w_s^{P(1,1)} = w_s^{P(0,0)}; \text{ if } e_o - \theta_o k_o > \chi, q_c^{P(1,1)} > q_c^{P(0,0)}. \\ where \chi = \frac{1}{p_e} [\sqrt{8p[c + p_e(e_c - \theta_c k_c)]} - 2p_e(e_c - \theta_c k_c) - p(1 + \delta_o \sqrt{k_o}) - 3c]. \end{array}$

When $\frac{w_s + p_e(e_o - \theta_o k_o)}{p} \ge \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}}$ is satisfied, it indicates that the profit per unit of OEM products is less than that of CM products. If $\delta_o \sqrt{k_o} + \frac{p_e(e_o - \theta_o k_o) + c}{n} \ge 1$ is satisfied, then the carbon emissions per unit of OEM products are relatively high, further reducing the profit margin of OEM products. To ensure its own profit, the OEM will produce at the minimum quantity, while TS sets the wholesale price to the maximum. Since the unmet demand by the OEM will be entirely transferred to the CM, the optimal production quantity for the CM increases. If $\delta_o \sqrt{k_o} + \frac{p_e(e_o - \theta_o k_o) + c}{p} < 1$ is satisfied, then the carbon emissions per unit of OEM products are relatively low, giving OEM products some profit margin. However, since the profit per unit of OEM products is still less than that of CM

products, the OEM will not produce excessively to completely substitute for the OEM's products. To ensure its own profit, the OEM will produce at the (0, 0) level, leading TS to also adopt the wholesale price under the (0, 0) condition. For CM, its production quantity will only increase when the carbon emissions per unit of OEM products are high, as the increased carbon emissions make CM's complete substitution of OEM's demand more likely. The results of Proposition 8 indicate that when the products of two manufacturers are completely substitutable, the optimal decisions of all members are not necessarily worse than the optimal decisions when no substitution exists.

Proposition 9. In the scenario of a purely competitive mode under (1, 1) conditions,

(1) When $\frac{w_s + p_e(e_o - \theta_o k_o)}{p} \leq \frac{1}{2} [\frac{c + p_e(e_c - \theta_c k_c)}{p}]^2$, $w_s^{P(1,1)}$ increases with the increase in both k_o and p_e , while decreasing with the increase in k_c ; $q_o^{P(1,1)}$ increases with the increase in k_o and decreases with the increase in k_c ; if $\frac{e_c - \theta_c k_c}{e_o - \theta_o k_o} > \frac{2\sqrt{2p[w_s^{P(1,1)} + p_e(e_o - \theta_o k_o)]}}{w_s^{P(1,1)} + p_e(e_o - \theta_o k_o) + c}$, $q_o^{P(1,1)}$ increases with the increase in p_e ; $q_c^{P(1,1)}$ increases with the increase in k_c and decreases with the increase in p_e , remaining unchanged with the variation in k_o ;

(2) When $\frac{w_s + p_e(e_o - \theta_o k_o)}{p} \ge \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}}$, if $\delta_o \sqrt{k_o} + \frac{p_e(e_o - \theta_o k_o) + c}{p} < 1$, $q_o^{P(1,1)}$ and $w_s^{P(1,1)}$ increases with the increase in both k_o and decreases with the increase in p_e , remaining unchanged with the variation in k_c ; $q_c^{P(1,1)}$ increases with the increase in k_c ; if $k_o < (\frac{p\delta_o}{2p_e\theta_o})^2$, $q_c^{P(1,1)}$ increases with the increase in k_o ; if $p_e > \frac{p}{2(e_o - \theta_o k_o)^2} - \frac{c}{e_c - \theta_c k_c}$, $q_c^{P(1,1)}$ increases with the increase in the increase in p_e ;

(3) When $\frac{w_s + p_e(e_o - \theta_o k_o)}{p} \ge \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}}$, if $\delta_o \sqrt{k_o} + \frac{p_e(e_o - \theta_o k_o) + c}{p} \ge 1$, $w_s^{P(1,1)}$ increases with the increase in p_e , remaining unchanged with the variation in k_c ; $q_o^{P(1,1)}$ increases with the increase in both k_o , remaining unchanged with the variation in k_c and p_e ; $q_c^{P(1,1)}$ increases with the increase in k_c and decreases with the increase in p_e , remaining unchanged with the variation in k_c and p_e ; $q_c^{P(1,1)}$ increases with the increase in k_c and decreases with the increase in p_e , remaining unchanged with the variation in k_o .

From Proposition 9, we observe that an increase in manufacturer's green investments leads to an increase in their yields. This is because manufacturer's green investments can bring about an increase in demand. Additionally, the increase in OEM green investments promotes TS to raise wholesale prices. This is because the increase in OEM green investments brings more demand to TS. To enhance their profits, TS raises wholesale prices. However, the impact of manufacturers' green investments on competitors varies with the situation. When $\frac{w_s + p_e(e_o - \theta_o k_o)}{p} \leq \frac{1}{2} \left[\frac{c + p_e(e_c - \theta_c k_c)}{p}\right]^2$, an increase in CM's green investments increases its yields, satisfying more demand. Conversely, OEM's demand decreases, resulting in a reduction in its production. To ensure its profit, TS lowers wholesale prices to encourage OEMs to increase production. Furthermore, since OEM's yields are large enough, there is no excess demand. Therefore, CM's production remains unchanged and does not vary with k_o . When $\frac{w_s + p_e(e_o - \theta_o k_o)}{p} \ge \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}}$ and $\delta_o \sqrt{k_o} + \frac{p_e(e_o - \theta_o k_o) + c}{p} < 1$, CM's yields are large enough, and there is no excess demand. Therefore, the optimal decisions of OEM and TS do not vary with k_c . However, only when OEM's green investment is low do CM's optimal yields increase with the increase of k_o . This is because when OEM's green investment is low, there is more excess demand. Thus, CM increases yields to satisfy larger substitution demand. When $\delta_o \sqrt{k_o} + \frac{p_e(e_o - \theta_o k_o) + c}{p} \ge 1$, OEM's unit production cost is high, so it will produce at the minimum level. At this time, CM increases yields to fully satisfy OEM's uncertain demand. Therefore, the yields of CM (or OEM) only increase with the increase in CM (or OEM) green investments.

Furthermore, we find that an increase in carbon trading prices always leads to a decrease in wholesale prices. This is because the increase in carbon trading prices increases the unit production cost of OEM products. To prevent significant yield cuts by the OEM, TS

lowers wholesale prices. Additionally, when the manufacturer's yields are sufficiently large, the increase in carbon trading prices significantly raises the cost of products, leading to a reduction in their yields. However, if yields are not sufficient, an increase in carbon trading prices can promote an increase in yields under specific circumstances. This phenomenon also illustrates that an increase in carbon trading prices does not necessarily suppress the yields of manufacturing enterprises.

5. Model Analysis of Co-Opetitive Supply Chain

In this section, we consider the scenario where OEM and CM are in a co-opetitive relationship. In this scenario, CM will become a supplier to OEM and compete with OEM in the final market. The supply chain structure under co-opetitive mode is illustrated in Figure 3.



Figure 3. Supply chain structure under co-opetitive mode.

At the beginning of the period, CM announces the wholesale price of raw materials, denoted as w_c . Then, OEM and CM, respectively, conduct their green investments, k_o and k_c , and determine their yields q_o and q_c . Meanwhile, OEM orders raw materials from CM and pays w_cq_o for CM. After the green investments, the unit carbon emissions of the final products of OEM and CM decrease to $(e_o - \theta_o k_0)$ and $(e_c - \theta_c k_c)$, respectively. At the end of the period, the final products of OEM and CM are sold at price p, and they, respectively, obtain carbon trading profits of $p_e[E_o - (e_o - \theta_o k_o)q_o]$ and $p_e[E_c - (e_c - \theta_c k_c)q_c]$ through carbon cap-and-trade system. The sequence of events is illustrated in Figure 4.



Figure 4. Sequence of events in the co-opetitive mode.

On this basis, we will analyze the co-opetitive mode under different product substitution scenarios.

5.1. Co-Opetitive Mode: Non-substitutable (0, 0)

First, we consider the scenario where there is no substitutability between the final products of OEM and CM. In this case, the profit functions of OEM and CM are, respectively, given by:

$$\pi_o^C = pE[\min[D_o, q_o]] - w_c q_o + p_e[E_o - (e_o - \theta_o k_o)q_o] - k_o$$
(13)

$$\pi_c^{\mathsf{C}} = pE[\min[D_c, q_c]] - cq_c + p_e[E_c - (e_c - \theta_c k_c)q_c] - k_c + (w_c - c)q_o \tag{14}$$

We solve this problem using backward induction and in Proposition 9, comparing it with the optimal solution of the (0, 0) scenario under a purely competitive mode.

Proposition 10. In the (0, 0) scenario,
$$w_c^{C(0,0)} = w_s^{P(0,0)}$$
, $q_c^{C(0,0)} = q_c^{P(0,0)}$, $q_o^{C(0,0)} = q_o^{P(0,0)}$, $\pi_o^{C(0,0)} = \pi_s^{P(0,0)}$, $\pi_c^{C(0,0)} = \pi_s^{P(0,0)} + \pi_c^{P(0,0)}$.

Proposition 10 reveals that in the (0, 0) scenario, whether in a purely competitive or co-opetitive mode, the optimal solutions and OEM's optimal profit remain unchanged. Despite CM assuming the role of supplier, OEM and CM operate independently of each other, so decisions made by one manufacture do not influence the other. Therefore, the optimal solutions remain constant in both scenarios. Since the optimal yields for OEM remain unchanged, OEM's profit remains unaffected. In the co-opetitive mode, where CM acts as the supplier to OEM, CM's optimal profit equals the sum of CM's profit with TS in the purely competitive mode.

5.2. Co-Opetitive Mode: OEM Substitutable for CM (r, 0)

When OEM products partially substitute for CM products, the profit functions for OEM and CM, respectively, are:

$$\pi_o^C = pE[\min[D_o + r(D_c - q_c)^+, q_o] - w_c q_o + p_e[E_o - (e_o - \theta_o k_o)q_o] - k_o$$
(15)

$$\pi_c^C = pE[\min[D_c, q_c]] - cq_c + p_e[E_c - (e_c - \theta_c k_c)q_c] - k_c + (w_c - c)q_o$$
(16)

In Proposition 11, we demonstrate a comparison between this optimal solution and the optimal solution for the (r, 0) scenario under a purely competitive mode.

Proposition 11. In the (r, 0) scenario, $w_c^{C(r,0)} = w_s^{P(r,0)}$, $q_c^{C(r,0)} = q_c^{P(r,0)}$, $q_o^{C(r,0)} = q_o^{P(r,0)}$, $\pi_o^{C(r,0)} = \pi_s^{P(r,0)} + \pi_c^{P(r,0)}$.

Similar to Proposition 10, Proposition 11 also indicates that in the (r, 0) scenario, both under purely competitive and co-opetitive modes, the optimal solution and the optimal profit for OEM remain unchanged. This is because in the co-opetitive mode, CM's products cannot substitute for OEM's products, so yields and wholesale price decisions remain independent of each other. Additionally, in the co-opetitive mode, when OEM makes optimal ordering quantity decisions, it only needs to consider the overflow of CM demand. Therefore, in the (r, 0) scenario, both modes exhibit the same optimal decisions. Since OEM's optimal production quantity remains unchanged, its profit remains unaffected. Furthermore, in the co-opetitive mode, where CM acts as a supplier to OEM, CM's optimal profit will be equal to the sum of CM's profit under purely competitive modes and TS's profit.

Corollary 1. When CM cannot substitute for OEM, the total profit and carbon emissions in the supply chain are the same under both co-opetitive and purely competitive modes.

Corollary 1 indicates that when CM's products cannot substitute for OEM's products, collaborating with CM or not will not affect the overall performance of the supply chain. Moreover, in this scenario, both modes have an equal impact on carbon emission reduction.

5.3. Co-Opetitive Mode: CM Substitutable for OEM (0, r)

When CM's products partially substitute for OEM's products, the profit functions for OEM and CM, respectively, are:

$$\pi_o^{\rm C} = pE[\min[D_o, q_o]] - w_s q_o + p_e[E_o - (e_o - \theta_o k_o)q_o] - k_o$$
(17)

$$\pi_c^C = pE[\min[D_c + r(D_o - q_o)^+, q_c] - cq_c + p_e[E_c - (e_c - \theta_c k_c)q_c] - k_c + (w_c - c)q_o$$
(18)

We define $G = \frac{\partial \pi_c}{\partial w_c}$, and provide the conditions for the optimal solutions of OEM and CM in Lemma 6.

Lemma 6. In the case of (0, r) in the co-opetitive mode,
(1) When
$$e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c-\theta_c k_c)]}{rp}} \le U$$
, if $e_c - \theta_c k_c > V$,
 $q_o^{C(0,r)} = \frac{(1-r)[p(1+\delta_o\sqrt{k_o})-c]+rp_e(e_c-\theta_c k_c)-p_e(e_o-\theta_o k_o)}{p(2-r)}$

$$q_{c}^{C(0,r)} = 1 + \delta_{c}\sqrt{k_{c}} + \frac{r[p(1+\delta_{o}\sqrt{k_{o}}) + (1-r)c - rp_{e}(e_{c} - \theta_{c}k_{c}) + p_{e}(e_{o} - \theta_{o}k_{o})]}{p(2-r)} - \sqrt{\frac{2r[c + p_{e}(e_{c} - \theta_{c}k_{c})]}{p}}{w_{c}^{C(0,r)}} = \frac{p(1+\delta_{o}\sqrt{k_{o}}) + (1-r)c - (1-r)p_{e}(e_{o} - \theta_{o}k_{o}) - rp_{e}(e_{c} - \theta_{c}k_{c})}{2-r}$$

$$(2) When \ e_{c} - \theta_{c}k_{c} + \frac{p(2-r)}{rp_{e}}\sqrt{\frac{2[c+p_{e}(e_{c} - \theta_{c}k_{c})]}{rp}} \leq U, \ if \ e_{c} - \theta_{c}k_{c} \leq V, \ q_{o}^{C(0,r)} = \delta_{o}\sqrt{k_{o}},$$

$$q_{c}^{C(0,r)} = 1 + \delta_{c}\sqrt{k_{c}} + r - \sqrt{\frac{2r[c+p_{e}(e_{c}-\theta_{c}k_{c})]}{p}}, w_{c}^{C(0,r)} = p - p_{e}(e_{o}-\theta_{o}k_{o});$$
(3) When $e_{c} - \theta_{c}k_{c} + \frac{p(2-r)}{rp_{e}}\sqrt{\frac{2[c+p_{e}(e_{c}-\theta_{c}k_{c})]}{rp}} > U$, if $q_{o}^{C(0,r)} - \delta_{o}\sqrt{k_{o}} + \frac{q_{c}^{C(0,r)} - \delta_{c}\sqrt{k_{o}}}{r} > 1$
the optimal wholesale price satisfies $G(w_{c}^{C(0,r)}) = 0$, and the optimal yields, respectively, satisfy:

$$q_o^{C(0,r)} = 1 + \delta_o \sqrt{k_o} - \frac{w_c^{C(0,r)} + p_e(e_o - \theta_o k_o)}{p}$$

$$q_{c}^{C(0,r)} = 1 + \delta_{c}\sqrt{k_{c}} - \frac{c + p_{e}(e_{c} - \theta_{c}k_{c})}{p} + \frac{r}{2}\left[\frac{w_{c}^{C(0,r)} + p_{e}(e_{o} - \theta_{o}k_{o})}{p}\right]^{2}$$

(4) When $e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c-\theta_c k_c)]}{rp}} \leq U$, if $q_o^{C(0,r)} - \delta_o \sqrt{k_o} + \frac{q_c^{C(0,r)} - \delta_c \sqrt{k_o}}{r} \leq 1$, the optimal wholesale price satisfies $G(w_c^{C(0,r)}) = 0$, and the optimal yields, respectively, satisfy:

$$q_o^{C(0,r)} = 1 + \delta_o \sqrt{k_o} - \frac{w_c^{C(0,r)} + p_e(e_o - \theta_o k_o)}{p}$$

$$q_{c}^{C(0,r)} = \sqrt{r^{2} \left[1 - \frac{w_{c}^{C(0,r)} + p_{e}(e_{o} - \theta_{o}k_{o})}{p}\right]^{2} + 2r \left[1 - \frac{c + p_{e}(e_{c} - \theta_{c}k_{c})}{p}\right] - r \left[1 - \frac{w_{c}^{C(0,r)} + p_{e}(e_{o} - \theta_{o}k_{o})}{p}\right] + \delta_{c}\sqrt{k_{c}}}$$

$$where \ U = \frac{1}{rp_{e}p} \left[p(1 + \delta_{o}\sqrt{k_{o}}) + (1 - r)c + p_{e}(e_{o} - \theta_{o}k_{o})\right], V = \frac{1}{rp_{e}} \left[p\delta_{o}\sqrt{k_{o}} - (1 - r)(p - c) + p_{e}(e_{o} - \theta_{o}k_{o})\right].$$

 $\begin{aligned} & \text{Proposition 12. In the } (0, r) \text{ scenario, when } e_c - \theta_c k_c \leq \min[H_1, U - \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c - \theta_c k_c)]}{rp}}], \\ & (1) \text{ When } e_c - \theta_c k_c > V, \text{ if } e_c - \theta_c k_c < \alpha, w_c^{C(0,r)} > w_s^{P(0,r)}; \text{ if } e_c - \theta_c k_c > \beta, q_o^{C(0,r)} > \\ & q_o^{P(0,r)}; \text{ if } e_c - \theta_c k_c < \gamma, q_c^{C(0,r)} > q_c^{P(0,r)}; \\ & (2) \text{ When } e_c - \theta_c k_c \leq V, w_c^{C(0,r)} > w_s^{P(0,r)}, q_o^{C(0,r)} < q_o^{P(0,r)}; \text{ if } e_o - \theta_o k_o > \frac{p(1 - \delta_o \sqrt{k_o}) - c}{p_e}, \\ & q_c^{C(0,r)} > q_c^{P(0,r)}. \\ & \text{where } \alpha = \frac{1}{2p_e} [p(1 + \delta_o \sqrt{k_o}) + c - p_e(e_o - \theta_o k_o)], \beta = \frac{1}{2p_e} [p(1 + \delta_o \sqrt{k_o}) - c + p_e(e_o - \theta_o k_o)]. \end{aligned}$

Since the optimal solutions for the three scenarios in the co-opetitive mode are implicit functions, Proposition 12 only compares the optimal solutions when the carbon emissions of CM products are lower. From Proposition 12, we observe that even with a reduction in CM carbon emissions, the comparison between the optimal decisions in the co-opetitive mode and the purely competitive mode still varies with the carbon emissions of CM and OEM unit products.

Proposition 13. *Comparing the* (0, *r*) *and* (0, 0) *scenarios in the co-opetitive mode, we obtain:*

(1) When $e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c-\theta_c k_c)]}{rp}} \le U$ and $e_c - \theta_c k_c > V$ are satisfied, if $e_o - \theta_o k_o > \varphi, q_c^{C(0,r)} > q_c^{C(0,0)}; \text{ if } e_c - \theta_c k_c > \kappa, q_o^{C(0,r)} > q_o^{C(0,0)}; \text{ if } e_o - \theta_o k_o < \varsigma, w_c^{C(0,r)} > Q_c^{C(0,r)} > Q_c^{C(0,r)}$ $w_{c}^{C(0,0)}$:

 $\begin{array}{l} w_{c} & , \\ (2) \text{ When } e_{c} - \theta_{c}k_{c} + \frac{p(2-r)}{rp_{e}} \sqrt{\frac{2[c+p_{e}(e_{c}-\theta_{c}k_{c})]}{rp}} \leq U \text{ and } e_{c} - \theta_{c}k_{c} \leq V \text{ are satisfied, } q_{c}^{C(0,r)} > \\ q_{c}^{C(0,0)}, q_{o}^{C(0,r)} \leq q_{o}^{C(0,0)}, w_{c}^{C(0,r)} \geq w_{c}^{C(0,0)}. \\ where \ \varphi = \frac{1}{p_{e}} [(2-r)[\sqrt{\frac{2p[c+p_{e}(e_{c}-\theta_{c}k_{c})]}{r}} - \frac{c+p_{e}(e_{c}-\theta_{c}k_{c})}{r}] - p(1+\delta_{o}\sqrt{k_{o}}) - (1-r)c + rp_{e}(e_{c}-\theta_{c}k_{c})], \\ \kappa = \frac{1}{2p_{e}} [p(1+\delta_{o}\sqrt{k_{o}}) - c + p_{e}(e_{o}-\theta_{o}k_{o})], \\ \varsigma = \frac{1}{p_{e}} [p(1+\delta_{o}\sqrt{k_{o}}) - c - 2p_{e}(e_{c}-\theta_{c}k_{c})]. \end{array}$

From the results of Proposition 13, we observe that when the carbon emissions per unit of CM products are relatively high ($e_c - \theta_c k_c > V$), the optimal decisions in the (0, r) scenario are favorable under specific conditions compared to the optimal decisions in the (0, 0) scenario. This indicates that increasing production quantity is advantageous when the competitor's carbon emissions per unit are high. However, when the carbon emissions per unit of CM products are relatively low ($e_c - \theta_c k_c \leq V$), the optimal decision for CM in the (0, r) scenario is better, but the optimal decision for OEM is better in the (0, 0) scenario. This suggests that the carbon emissions per unit of both products have a significant impact on optimal decisions. Moreover, when the CM can substitute for excess demand from the OEM, it does not necessarily harm the optimal decisions.

Proposition 14. In the co-opetitive mode with the (0, r) scenario,

(1) When $e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c - \theta_c k_c)]}{rp}} \le U$ and $e_c - \theta_c k_c > V$, if $e_c - \theta_c k_c > \beta$, $w_c^{C(0,r)}$ decreases with the increase in r, $q_o^{C(0,r)}$ increases with the increase in r; if $\frac{2}{p(2-r)^2} [p(1+r)]$ $\delta_o \sqrt{k_o}) + p_e(e_o - \theta_o k_o)] + \frac{c}{p} > \frac{r(4-r)}{p(2-r)^2} p_e(e_c - \theta_c k_c) + \sqrt{\frac{c+p_e(e_c - \theta_c k_c)}{rp}}, q_c^{C(0,r)}$ increases with the increase in r;

(2) When $e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c - \theta_c k_c)]}{rp}} \le U$ and $e_c - \theta_c k_c \le V$, $w_c^{C(0,r)}$ and $q_o^{C(0,r)}$ do not change with the variation in r; if $r > \frac{c + p_e(e_c - \theta_c k_c)}{2p}$, $q_c^{C(0,r)}$ increases with the increase in r;

(3) When $e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c - \theta_c k_c)]}{rp}} > U$ and $q_o^{C(0,r)} - \delta_o \sqrt{k_o} + \frac{q_c^{C(0,r)} - \delta_c \sqrt{k_o}}{r} > 1$, $w_c^{C(0,r)}$ and $q_o^{C(0,r)}$ increases with the increase in r, while $q_c^{C(0,r)}$ decreases with the increase in r.

Proposition 14 demonstrates that the impact of the substitution rate r on the optimal decisions varies under different conditions. This differs from the results presented in Proposition 6. This is because CM acts as the supplier to OEM. When CM can substitute for the excess demand of OEM, CM can influence OEM's decisions by controlling the wholesale price. Therefore, in the co-opetitive mode, the effect of the substitution rate r on the optimal decisions differs from that in the purely competitive mode.

Proposition 15. *In the co-opetitive mode with the* (0, *r*) *scenario,*

(1) When $e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c-\theta_c k_c)]}{rp}} \leq U$ and $e_c - \theta_c k_c > V$, $w_c^{C(0,r)}$ increases with the increase in k_c and k_o , and decreases with the increase in p_e ; $q_o^{C(0,r)}$ increases with the increase in k_o ; and decreases with the increase in k_c ; if $r > \frac{e_o - \theta_o k_o}{e_c - \theta_c k_c}$, $q_o^{C(0,r)}$ increases with the increase in p_e ; $q_c^{C(0,r)}$ increases with the increase in k_c ; if $k_o < (\frac{p\delta_o}{2p_c\theta_o})^2$, $q_c^{C(0,r)}$ increases with the increase in k_o ; if $p_e > \frac{p(2-r)^2(e_c-\theta_ck_c)}{2[(e_o-\theta_ok_o)-r(e_c-\theta_ck_c)]^2} - \frac{c}{e_c-\theta_ck_c}$, $q_c^{C(0,r)}$ increases with the increase in p_e ; (2) When $e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c-\theta_ck_c)]}{rp}} \le U$ and $e_c - \theta_c k_c \le V$, $w_c^{C(0,r)}$ increases with the increase in k_o and decreases with the increase in p_e , but remains unchanged with the

(2) When $e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c - v_c \kappa_c)]}{rp}} \leq U$ and $e_c - \theta_c k_c \leq V$, $w_c^{C(0,r)}$ increases with the increase in k_o and decreases with the increase in p_e , but remains unchanged with the increase in k_c ; $q_o^{C(0,r)}$ increases with the increase in k_o , but remains unchanged with the increase in k_c and p_e ; $q_c^{C(0,r)}$ increases with the increase in k_c and decreases with the increase in k_c ;

(3) When
$$e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c-\theta_c k_c)]}{rp}} > U$$
 and $q_o^{C(0,r)} - \delta_o \sqrt{k_o} + \frac{q_c^{C(0,r)} - \delta_c \sqrt{k_o}}{r} > 1$,

 $w_c^{C(0,r)} \text{ decreases with the increase in } p_e \text{ and remains unchanged with the increase in } k_c; \text{ if } \frac{\delta_o}{2\sqrt{k_o}p_e\theta_o} < \vartheta, w_c^{C(0,r)} \text{ increases with the increase in } k_o; q_o^{C(0,r)} \text{ decreases with the increase in } p_e \text{ and remains unchanged with the increase in } k_c; \text{ if } 2p_e\theta_o\sqrt{k_o} - \delta_op^2 < \frac{\delta_op^2}{\vartheta}, q_o^{C(0,r)} \text{ increases } with the increase in } k_c; \text{ if } 2p_e\theta_o\sqrt{k_o} - \delta_op^2 < \frac{\delta_op^2}{\vartheta}, q_o^{C(0,r)} \text{ increases } with the increase in } k_c; \text{ if } 2p_e\theta_o\sqrt{k_o} < \delta_op^2 < \vartheta, q_c^{C(0,r)} \text{ increases } with the increase in } k_c; \text{ if } \frac{\delta_op}{2p_e\theta_o\sqrt{k_o}} < \vartheta, q_c^{C(0,r)} \text{ increases } with the increase in } k_c; \text{ if } \frac{\delta_op}{2p_e\theta_o\sqrt{k_o}} < \vartheta, q_c^{C(0,r)} \text{ increases } with the increase in } k_c; \text{ if } w_c^{C(0,r)} + p_e(e_o - \theta_o k_o)](e_o - \theta_o k_o) > (e_c - \theta_c k_c)\vartheta, q_c^{C(0,r)} \text{ increases } with the increase in } p_e.$ $where \vartheta = \frac{3r^2}{2p^3}[w_c^{C(0,r)} + p_e(e_o - \theta_o k_o)]^2 + \frac{2}{p} - \frac{r^2}{p}[w_c^{C(0,r)} + p_e(e_o - \theta_o k_o)].$

From Proposition 15, we have identified a new phenomenon. In the case of $e_c - \theta_c k_c + \frac{p(2-r)}{rp_e} \sqrt{\frac{2[c+p_e(e_c-\theta_c k_c)]}{rp}} > U$ and $q_o^{C(0,r)} - \delta_o \sqrt{k_o} + \frac{q_c^{C(0,r)} - \delta_c \sqrt{k_o}}{r} > 1$, the increase in OEM's green investment does not necessarily lead to an increase in its production. This is because CM acts as OEM's supplier and can also substitute for OEM's excess demand. Therefore, under specific conditions, CM can make the OEM's green investment unable to improve the OEM's production by adjusting the wholesale price. Additionally, Proposition 13 also indicates that the increase in carbon trading prices does not necessarily suppress manufacturers' production.

5.4. Co-Opetitive Mode: CM and OEM Symmetrically Substitutable (r, r)

When CM and OEM products are symmetrically substitutable, the profit functions for OEM and CM are as follows:

$$\pi_o^C = pE[\min[D_o + r(D_c - q_c)^+, q_o] - w_s q_o + p_e[E_o - (e_o - \theta_o k_o)q_o] - k_o$$
(19)

 $\pi_c^C = pE[\min[D_c + r(D_o - q_o)^+, q_c] - cq_c + p_e[E_c - (e_c - \theta_c k_c)q_c] - k_c + (w_c - c)q_o$ (20)

Similar to Section 4.4, the complexity of this problem is exceptionally high. Therefore, to simplify the problem, we consider the special case of r = 1, where there is complete product substitution between OEM and CM. In this case, market demand that OEM (or CM) cannot satisfy will be entirely transferred to CM (or OEM). This situation also exists in reality. We provide the conditions for the optimal solutions of OEM and CM in Lemma 7.

Lemma 7. *In the* (1, 1) *scenario under the co-opetitive mode:*

(1) When $\frac{w_c + p_e(e_o - \theta_o k_o)}{p} \leq \frac{1}{2} \left[\frac{c + p_e(e_c - \theta_c k_c)}{p}\right]^2$, $q_c^{C(1,1)} = 1 + \delta_c \sqrt{k_c} - \frac{c + p_e(e_c - \theta_c k_c)}{p}$, the optimal wholesale price satisfies $G(w_c^{C(0,r)}) = 0$, and the optimal yields for OEM satisfy:

$$q_o^{C(1,1)} = 1 + \delta_o \sqrt{k_o} + \frac{c + p_e(e_c - \theta_c k_c)}{p} - \sqrt{\frac{2[w_c^{C(1,1)} + p_e(e_o - \theta_o k_o)]}{p}}$$

$$\begin{array}{ll} (2) \quad \text{When} \quad \frac{1}{2} [\frac{e+p_{c}(e_{c}-\theta_{c}k_{c})}{p}]^{2} &< \frac{w_{c}+p_{c}(e_{o}-\theta_{c}k_{c})}{p} &< \sqrt{\frac{2[c+p_{c}(e_{c}-\theta_{c}k_{c})]}{p}}, \quad \text{if}\\ q_{c} - \delta_{c}\sqrt{k_{c}} + q_{o} - \delta_{o}\sqrt{k_{o}} > 1, \text{ the optimal wholesale price satisfies } G(w_{c}^{C(0,r)}) = 0, \text{ and the optimal yields, respectively, satisfy:} \\ 8[1 + \delta_{c}\sqrt{k_{c}} - q_{c}^{C(1,1)} - \frac{c + p_{c}(e_{c} - \theta_{c}k_{c})}{p}] = -\{\frac{2[w_{c}^{C(1,1)} + p_{e}(e_{o} - \theta_{o}k_{o})]}{p} - [1 - (q_{c}^{C(1,1)} - \delta_{c}\sqrt{k_{c}})]^{2}\}^{2} \\ 8[1 + \delta_{o}\sqrt{k_{o}} - q_{o}^{C(1,1)} - \frac{w_{s}^{C(1,1)} + p_{e}(e_{o} - \theta_{o}k_{o})}{p}] = -\{\frac{2[c + p_{e}(e_{c} - \theta_{c}k_{c})]}{p} - [1 - (q_{o}^{C(1,1)} - \delta_{o}\sqrt{k_{o}})]^{2}\}^{2} \\ (3) \quad \text{When} \quad \frac{1}{2}[\frac{(e+p_{c}(e_{c} - \theta_{c}k_{c}))}{p}]^{2} < \frac{w_{c}+p_{c}(e_{o} - \theta_{o}k_{o})}{p} - [1 - (q_{o}^{C(1,1)} - \delta_{o}\sqrt{k_{o}})]^{2}\}^{2} \\ \delta_{o}\sqrt{k_{o}} \leq 1, \text{ the optimal wholesale price satisfies } G(w_{c}^{C(0,r)}) = 0, \text{ and the optimal yields, respectively, satisfy:} \\ \frac{q_{c}^{C(1,1)} - \delta_{c}\sqrt{k_{c}}}{2} + \frac{p - c - p_{e}(e_{c} - \theta_{c}k_{c})}{p(q_{c}^{C(1,1)} - \delta_{c}\sqrt{k_{c}})^{2}} = \sqrt{(q_{c}^{C(1,1)} - \delta_{c}\sqrt{k_{c}})^{2} + 2[1 - \frac{w_{c}^{C(1,1)} + p_{e}(e_{o} - \theta_{o}k_{o})}{p}]} \\ \frac{q_{0}^{C(1,1)} - \delta_{o}\sqrt{k_{o}}}{2} + \frac{p - w_{s}^{C(1,1)} - p_{c}(e_{a} - \theta_{o}k_{o})}{p(q_{o}^{C(1,1)} - \delta_{o}\sqrt{k_{o}})} = \sqrt{(q_{c}^{C(1,1)} - \delta_{o}\sqrt{k_{o}})^{2} + 2[1 - \frac{c + p_{c}(e_{c} - \theta_{c}k_{c})}{p}]} \\ \frac{q_{0}^{C(1,1)} - \delta_{o}\sqrt{k_{o}}}{p(q_{0}^{C(1,1)} - \delta_{o}\sqrt{k_{o}})} = \sqrt{(q_{o}^{C(1,1)} - \delta_{o}\sqrt{k_{o}})^{2} + 2[1 - \frac{c + p_{c}(e_{c} - \theta_{c}k_{c})}{p}]} \\ \frac{q_{0}^{C(1,1)} - \delta_{o}\sqrt{k_{o}}}{p(q_{o}^{C(1,1)} - \delta_{o}\sqrt{k_{o}}})} = \sqrt{(q_{o}^{C(1,1)} - \delta_{o}\sqrt{k_{o}})^{2} + 2[1 - \frac{c + p_{c}(e_{c} - \theta_{c}k_{c})}{p}]} \\ \frac{q_{0}^{C(1,1)} - \delta_{o}\sqrt{k_{o}}}}{p(q_{0}^{C(1,1)} - \delta_{o}\sqrt{k_{o}})} \geq \sqrt{2[e+p_{c}(e_{c} - \theta_{c}k_{o}]]}, \text{if }\delta_{o}\sqrt{k_{o}} \geq \frac{p_{c}(e_{c} - \theta_{c}k_{c})}{p}} \\ \frac{q_{0}^{C(1,1)} = 1 + \delta_{c}\sqrt{k_{c}}} - \sqrt{2[e+p_{c}(e_{c} - \theta_{c}k_{c})]}}{p(q_{o}^{C(1,1)} - \delta_{o}\sqrt{k_{o}}} + \frac{p_{c}(e_{c} - \theta_{c}k_{c})}{p}$$

Proposition 16. *In the* (1, 1) *scenario under the co-opetitive mode:*

 $(1) When both modes satisfy \frac{w+p_e(e_o-\theta_o k_o)}{p} \leq \frac{1}{2} [\frac{c+p_e(e_c-\theta_c k_c)}{p}]^2, w_c^{C(1,1)} = w_s^{P(1,1)}, q_c^{C(1,1)} = q_c^{P(1,1)}, \pi_o^{C(1,1)} = \pi_o^{P(1,1)}, \pi_c^{C(1,1)} = \pi_c^{P(1,1)} + \pi_s^{P(1,1)};$ $(2) When both modes satisfy \frac{w+p_e(e_o-\theta_o k_o)}{p} \geq \sqrt{\frac{2[c+p_e(e_c-\theta_c k_c)]}{p}} and \delta_o \sqrt{k_o} \geq \max[\frac{p_e(e_c-\theta_c k_c)}{p}, 1 - \frac{p_e(e_o-\theta_o k_o)+c}{p}], w_c^{C(1,1)} = w_s^{P(1,1)}, q_c^{C(1,1)} = q_c^{P(1,1)}, q_o^{C(1,1)} = q_o^{P(1,1)}, \pi_o^{C(1,1)} = \pi_o^{P(1,1)}, \pi_c^{C(1,1)} = \pi_o^{P(1,1)}, \pi_c^{P(1,1)} = \pi_o^{P(1,1)}, \pi_c^{$

(3) When both modes satisfy $\frac{w+p_e(e_o-\theta_o k_o)}{p} \ge \sqrt{\frac{2[c+p_e(e_c-\theta_c k_c)]}{p}}$ and $\delta_o \sqrt{k_o} < \min[\frac{p_e(e_c-\theta_c k_c)}{p}]$, $1 - \frac{p_e(e_o-\theta_o k_o)+c}{p}]$, $q_c^{C(1,1)} + q_o^{C(1,1)} = q_c^{P(1,1)} + q_o^{P(1,1)}$; if $e_c - \theta_c k_c < \frac{1}{2p_e}[p(1+\delta_o \sqrt{k_o}) - c + Q(1,1)] = Q(1,1)$ $p_e(e_o - \theta_o k_o)], w_c^{C(1,1)} > w_s^{P(1,1)}, q_c^{C(1,1)} > q_c^{P(1,1)}, q_o^{C(1,1)} < q_o^{P(1,1)}.$

Due to the complexity of the (1, 1) scenario, Proposition 16 only allows for partial comparisons between the two modes. From Proposition 16, it is evident that when both modes satisfy $\frac{w+p_e(e_o-\theta_o k_o)}{p} \leq \frac{1}{2} [\frac{c+p_e(e_c-\theta_c k_c)}{p}]^2$, there is no difference in the optimal solutions between the two modes. This is because the OEM's production is sufficiently large, eliminating any excess demand. Even with higher wholesale prices in the co-opetitive mode, CM cannot attract additional demand. Consequently, the profits of the OEM remain the same under both modes, and in the co-opetitive mode, CM sets its optimal decision to be

the same as in the purely competitive mode. Additionally, as CM serves as the OEM's supplier in the co-opetitive mode, CM's profits in this mode are the sum of those in the purely competitive mode and the profits of the TS. Furthermore, since the optimal decisions are the same under both modes, the carbon emissions are also identical at this point. When both modes satisfy $\frac{w+p_e(e_o-\theta_o k_o)}{p} \ge \sqrt{\frac{2[c+p_e(e_c-\theta_c k_c)]}{p}}$, if $\delta_o \sqrt{k_o} < \min[\frac{p_e(e_c-\theta_c k_c)}{p}, 1-\frac{p_e(e_o-\theta_o k_o)+c}{p}]$, the OEM will produce at the minimum capacity. In this scenario, all uncertain demands of the OEM cannot be satisfied. Consequently, CM will increase its yields to fulfill all uncertain demands of the OEM. Additionally, as the supplier to the OEM, CM sets the wholesale price to maximize its revenue. Therefore, the optimal decisions are the same under both modes, and in the co-opetitive mode, CM's profit is the sum of the profit in the purely competitive mode and that of the TS. However, if $\delta_o \sqrt{k_o} < \min[\frac{p_e(e_c - \theta_c k_c)}{p}, 1 - \frac{p_e(e_o - \theta_o k_o) + c}{p}]$, the relationship between the optimal decisions under both modes is influenced by the CM's unit product carbon emissions. This is because, in the co-opetitive mode, CM can adjust its demand by controlling the wholesale price, effectively mitigating the risk of greater losses from excessive production when the CM's unit product carbon emissions are high. Furthermore, we observe that the total yields of both manufacturers remain unchanged under both modes. This phenomenon indicates that the market satisfaction rate remains constant regardless of the mode chosen.

Proposition 17. Comparing the (1, 1) and (0, 0) scenarios in the co-opetitive mode, we obtain:

(1) When $\frac{w_c + p_e(e_o - \theta_o k_o)}{p} \ge \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}}$ and $\delta_o \sqrt{k_o} < \frac{p_e(e_c - \theta_c k_c)}{p}$ are satisfied, if $e_o - \theta_o k_o > \psi$, $q_c^{C(1,1)} > q_c^{C(0,0)}$; if $e_c - \theta_c k_c > \iota$, $q_o^{C(1,1)} > q_o^{C(0,0)}$, if $e_o - \theta_o k_o < v$, $w_c^{C(1,1)} > w_c^{C(0,0)}$;

 $\begin{array}{l} w_{c}^{(1)} + y \geq w_{c}^{(1)} + y_{c}^{(1)} \\ (2) \ When \ \frac{w_{c} + p_{e}(e_{o} - \theta_{o}k_{o})}{p} \geq \sqrt{\frac{2[c + p_{e}(e_{c} - \theta_{c}k_{c})]}{p}} \ and \ \delta_{o}\sqrt{k_{o}} \geq \frac{p_{e}(e_{c} - \theta_{c}k_{c})}{p} \ are \ satisfied, \\ q_{c}^{(1,1)} > q_{c}^{(0,0)}, \ q_{o}^{(1,1)} \leq q_{o}^{(0,0)}, \ w_{c}^{(1,1)} \geq w_{c}^{(0,0)}. \\ where \ \psi = \frac{1}{p_{e}} [\sqrt{2p[c + p_{e}(e_{c} - \theta_{c}k_{c})]} - p(1 + \delta_{o}\sqrt{k_{o}}) - c], \ [\iota = \frac{1}{2p_{e}} [p(1 + \delta_{o}\sqrt{k_{o}}) - c + \frac{1}{2p_{e}} [p(1 + \delta_{o}\sqrt{k_{o}}) - c]. \end{array}$ $p_e(e_o- heta_ok_o)$], $v=rac{1}{p_e}[p(1+\delta_o\sqrt{k_o})-c-2p_e(e_c- heta_ck_c)].$

When $\left[\frac{w_c + p_e(e_o - \theta_o k_o)}{p} \ge \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}}\right]$ is satisfied, it indicates that the profit per unit of OEM products is less than that of CM products. From Proposition 17, we observe that when the carbon emissions per unit of CM products are relatively high $(\delta_o \sqrt{k_o} < \frac{p_e(e_c - \theta_c k_c)}{p})$, both CM and OEM's optimal decisions in the (1, 1) scenario may be superior to the optimal decisions in the (0, 0) scenario. This result suggests that increasing production quantity is advantageous when the competitor's carbon emissions per unit are high. When the carbon emissions per unit of CM products are relatively low $(\delta_o \sqrt{k_o} \ge \frac{p_e(e_c - \Theta_c k_c)}{p})$, CM in the (1, 1) scenario can completely substitute for more of OEM's demand. To maximize profits, CM in the (1, 1) scenario will set the wholesale price to the maximum, leading OEM to produce at the minimum quantity. Therefore, the results of Proposition 17 indicate that when the products of two manufacturers are completely substitutable, the optimal decisions of all members are not necessarily worse than the optimal decisions when no substitution exists.

Proposition 18. *In the* (1, 1) *scenario under the co-opetitive mode:*

(1) When $\frac{w_c + p_e(e_o - \theta_o k_o)}{p} \leq \frac{1}{2} \left[\frac{c + p_e(e_c - \theta_c k_c)}{p}\right]^2$, $w_c^{C(1,1)}$ increases with the increase in k_o and p_e , and decreases with the increase in k_c ; $q_o^{C(1,1)}$ increases with the increase in k_o and k_c ; if $\frac{e_c - \theta_c k_c}{e_o - \theta_o k_o} > \frac{2\sqrt{2p[w_c^{C(1,1)} + p_e(e_o - \theta_o k_o)]}}{w_c^{C(1,1)} + p_e(e_o - \theta_o k_o) + c}, q_o^{C(1,1)} \text{ increases with the increase in } p_e; q_c^{C(1,1)} \text{ increases with the increase in } p_e; q_c^{C(1,1)} \text{ increases with the increase in } p_e; q_c^{C(1,1)} \text{ increases in } p_e;$

the increase in k_c and decreases with the increase in p_e , without varying with k_o ; (2) When $\frac{w_c + p_e(e_o - \theta_o k_o)}{p} \ge \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}}$ and $\delta_o \sqrt{k_o} < \frac{p_e(e_c - \theta_c k_c)}{p}$, $w_c^{C(1,1)}$ increases with the increase in p_e ; $q_c^{C(1,1)}$ increases with the

increase in k_c and decreases with the increase in k_o ; if $\frac{e_o - \theta_o k_o}{e_c - \theta_c k_c} > 1 + \frac{p}{\sqrt{2p[c + p_e(e_c - \theta_c k_c)]}}$, $q_c^{C(1,1)}$ increases with the increase in p_e ; $q_o^{C(1,1)}$ increases with the increase in k_o and decreases with the

increases with the increase in p_e ; $q_o^{C(1,r)}$ increases with the increase in k_o and decreases with the increase in k_c ; if $e_c - \theta_c k_c > e_o - \theta_o k_o$, $q_o^{C(1,1)}$ increases with the increase in p_e ; (3) When $\frac{w_c + p_e(e_o - \theta_o k_o)}{p} \ge \sqrt{\frac{2[c + p_e(e_c - \theta_c k_c)]}{p}}$ and $\delta_o \sqrt{k_o} \ge \frac{p_e(e_c - \theta_c k_c)}{p}$, $w_c^{C(1,1)}$ increases

with the increase in k_o and decreases with the increase in p_e , without changing with k_c ; $q_o^{C(1,1)}$ increases with the increase in k_o , without changing with k_c and p_e ; $q_c^{C(1,1)}$ increases with the increase in k_c and decreases with the increase in p_e , without changing with k_o .

Similarly, due to the complexity of the problem in the (1, 1) scenario, in Proposition 18, we can only demonstrate the influence of green investment and carbon trading prices on optimal decisions in some cases. The results of Proposition 18 also indicate that in the co-opetitive mode, when CM can replace OEM, carbon trading prices may not necessarily suppress the production of manufacturers.

6. Numerical Analysis

In this section, we continue to conduct a numerical analysis and explore new findings by consulting the carbon trading prices from the China Carbon Emission Trade Exchange (CCETE) and using partial parameter settings from ref. [11]. We assume that p is 5, c is 1, p_e is 4, r is 0.5, e_i (i = o, c) are both 1, θ_i (i = o, c) are both 0.1, δ_i (i = o, c) are both 0.01, and E_i (i = o, c) are both 10. Next, we set CM's green investment k_c to be 3.5 and 6.5, respectively, and use OEM's green investment k_o as the horizontal axis to compare and analyze the profits of OEM π_o , CM π_c , and the total profit π of the supply chain under different scenarios.

First, we compare the three scenarios under purely competitive: (0, 0), (r, 0), (0, r)and (1, 1), and present the corresponding results in Figures 5–7. From Figures 5 and 6, we observe that $\pi_o^{P(r,0)} = \pi_o^{P(0,0)}$ and $\pi_c^{P(0,r)} = \pi_c^{P(0,0)}$ always exist. This further validates the results analyzed in Propositions 2 and 5. Additionally, we observe that whether $k_c = 3.5$ or $k_c = 6.5$ is satisfied, when k_o is relatively small, $\pi_o^{P(r,0)}$ is always optimal. However, as k_o increases, $\pi_o^{P(1,1)}$ becomes optimal. This is because in the (r, 0) scenario of the purely competitive mode, the CM cannot substitute for the excess demand of the OEM, whereas in the (1, 1) scenario, the CM can fully substitute for the demand of the OEM. Therefore, when k_0 is small, the unmet demand by the OEM is significant, and in the (r, 0) scenario, this excess demand of the OEM does not transfer to the CM. In contrast, in the (1, 1) scenario, the excess demand of the OEM is fully transferred to the CM. At this point, the OEM's sales revenue in the (1, 1) scenario significantly decreases, leading to $\pi_0^{P(r,0)} > \pi_0^{P(1,1)}$. However, when k_o is large, the OEM can substitute for the unmet excess demand of the CM. In the (r, 0) scenario, the OEM can only partially substitute for the excess demand of the CM, while in the (1, 1) scenario, the OEM can fully substitute for the excess demand of the CM. At this point, the OEM can achieve higher sales revenue in the (1, 1) scenario, leading to $\pi_{0}^{P(1,1)} > \pi_{0}^{P(r,0)}.$



Figure 5. Comparison of π_o^p in the scenarios of (0, 0), (r, 0), (0, r) and (1, 1) under purely competitive mode. (a) $k_c = 3.5$. (b) $k_c = 6.5$.



Figure 6. Comparison of π_c^p in the scenarios of (0, 0), (r, 0), (0, r) and (1, 1) under purely competitive mode. (a) $k_c = 3.5$. (b) $k_c = 6.5$.



Figure 7. Comparison of π^p in the scenarios of (0, 0), (r, 0), (0, r) and (1, 1) under purely competitive mode. (a) $k_c = 3.5$. (b) $k_c = 6.5$.

From Figure 6, we observe that when $k_c = 3.5$ is satisfied, $\pi_c^{P(1,1)}$ is always optimal. However, when $k_c = 6.5$ is satisfied, if k_o is relatively small, $\pi_c^{P(1,1)}$ is no longer optimal. As k_o increases, $\pi_c^{P(1,1)}$ becomes optimal. This is because when k_c is small, in the (1, 1) scenario of the purely competitive mode, the CM can fully substitute for the excess demand of the OEM. The additional revenue CM gains from substitution can offset the carbon emission costs incurred during the production process. Therefore, $\pi_c^{P(1,1)}$ is optimal. However, when k_c is large and k_o is small, in the (1, 1) scenario, the CM can fully substitute for more excess demand of the OEM. At this point, the additional revenue from substitution does not offset the carbon emission costs incurred during the production process, making $\pi_c^{P(1,1)}$ not optimal. As k_o increases, the excess demand of the OEM decreases, and the additional revenue CM gains from substitution can again offset the carbon emission costs incurred during the production process. Thus, $\pi_c^{P(1,1)}$ becomes optimal again.

For the entire supply chain, we observe from Figure 7 that when k_c is relatively small, $\pi^{P(1,1)}$ is always optimal. However, when k_c is large, if k_o is small, $\pi^{P(1,1)}$ is no longer optimal. As k_o increases, $\pi^{P(1,1)}$ becomes optimal. This is because when k_c is small, in the (1, 1) scenario of the purely competitive mode, the additional revenue CM gains from substitution can not only offset its own carbon emission costs incurred during the production process but also compensate for the losses of OEM and TS. Therefore, $\pi^{P(1,1)}$ is optimal. However, when k_c is large and k_o is small, the additional revenue CM gains from substitution in the (1, 1) scenario cannot offset its own carbon emission costs or the losses incurred by OEM and TS, making $\pi^{P(1,1)}$ not optimal. As k_o increases, the unmet excess demand of OEM decreases, and at this point, the additional revenue CM gains from substitution can again offset its own carbon emission costs and compensate for the losses incurred by OEM and TS. Therefore, $\pi^{P(1,1)}$ becomes optimal again. This result indicates that in the purely competitive mode, product substitution can bring better profits to the supply chain.

Next, we compare the co-opetitive mode under scenarios (0, 0), (r, 0), (0, r) and (1, 1), and present the corresponding results in Figures 8–10. From Figure 8, we observe that regardless of the value of k_c , $\pi_o^{C(0,r)}$ is never optimal and consistently lower than $\pi_o^{C(0,0)}$. This is because, in the co-opetitive mode, CM serves as the manufacturer for OEM. When only CM can substitute for OEM's products, CM will set higher wholesale prices to suppress OEM's production, resulting in $\pi_o^{C(0,r)} < \pi_o^{C(0,0)}$. Propositions 9 and 10 indicate that in both scenarios (0, 0) and (r, 0), $\pi_o^{C(0,0)} = \pi_o^{P(0,0)}$ and $\pi_o^{C(r,0)} = \pi_o^{P(r,0)}$ always exist. Therefore, $\pi_o^{C(0,0)}$ and $\pi_o^{C(1,1)}$ yield the same results as in Figure 5. However, unlike Figure 5, when k_o is small, $\pi_o^{C(1,1)}$ is the minimum in the (1, 1) scenario under the co-opetitive mode. This is because, in the (1, 1) scenario, CM can not only fully substitute for OEM's excess demand but also manipulate the wholesale price, leading to a further decrease in OEM's production. Consequently, the sales revenue of OEM in the (1, 1) scenario significantly decreases, resulting in $\pi_o^{C(1,1)}$ being the minimum.



Figure 8. Comparison of π_o^C in the scenarios of (0, 0), (r, 0), (0, r) and (1, 1) under co-opetitive mode. (a) $k_c = 3.5$. (b) $k_c = 6.5$.



Figure 9. Comparison of π_c^C in the scenarios of (0, 0), (r, 0), (0, r) and (1, 1) under co-opetitive mode. (**a**) $k_c = 3.5$. (**b**) $k_c = 6.5$.



Figure 10. Comparison of π^{C} in the scenarios of (0, 0), (r, 0), (0, r) and (1, 1) under co-opetitive mode. (a) $k_{c} = 3.5$. (b) $k_{c} = 6.5$.

From Figure 9, we find that $\pi_c^{C(1,1)}$ is not necessarily optimal. Specifically, when $k_c = 6.5$ is satisfied, $\pi_c^{C(r,0)}$ is optimal. This is because when k_c is small, if k_0 is also small, the demand that the OEM cannot satisfy is high. In this case, under the (1, 1) scenario, the CM can fully substitute for the OEM's excess demand. However, the revenue gained by the CM through substitution cannot offset the carbon emission costs incurred during its production process. Therefore, $\pi_c^{C(1,1)}$ is not optimal. As k_o increases, the unmet demand by the OEM gradually decreases. In the (1, 1) scenario, the revenue gained by the CM through substitution can offset the carbon emission costs incurred during its production process, and as a supplier, the CM can further suppress the OEM's production by raising the wholesale price, making $\pi_c^{C(1,1)}$ optimal. However, as k_o further increases, the OEM's production significantly increases. In the (1, 1) scenario, the OEM can fully substitute for the CM's excess demand, leading to a decrease in the CM's sales revenue, making $\pi_c^{C(1,1)}$ not optimal. When k_c is large, in the (1, 1) scenario, the revenue gained by the CM through substitution can never offset the costs incurred from its green investments. Moreover, the CM, in an attempt to gain more excess demand, will raise the wholesale price to suppress the OEM's production. However, a sufficiently high wholesale price leads to a significant reduction in the OEM's production, resulting in a substantial decrease in the CM's profit from selling raw materials. Therefore, $\pi_c^{C(1,1)}$ is not optimal. Conversely, in the (r, 0) scenario, OEM can satisfy CM's excess demand, prompting OEM to increase its production. Consequently, CM earns more revenue by selling more raw materials, making $\pi_c^{C(r,0)}$ optimal.

In Figure 10, we similarly observe that $\pi^{C(1,1)}$ is not necessarily optimal. Specifically, when $k_c = 6.5$ is satisfied, $\pi^{C(r,0)}$ is optimal, for reasons similar to those in Figure 9. When k_c is small, if k_o is also small, the unmet demand from the OEM is high. In this case, under the (1, 1) scenario, the CM can fully substitute for the OEM's excess demand. However, the revenue gained by the CM through substitution cannot offset the carbon emission costs incurred during its production process. Therefore, $\pi^{C(1,1)}$ is not optimal. As k_0 increases, the unmet demand from the OEM gradually decreases. In the (1, 1) scenario, the revenue gained by the CM through substitution can offset the carbon emission costs incurred during its production process. Moreover, as a supplier, the CM can further suppress the OEM's production by raising the wholesale price, making $\pi^{C(1,1)}$ optimal. However, as k_0 further increases, the OEM's production significantly increases. In the (1, 1) scenario, the OEM can fully substitute for the CM's excess demand, leading to a decrease in the CM's sales revenue, making $\pi^{C(1,1)}$ not optimal. When k_c is large, in the (1, 1) scenario, the revenue gained by the CM through substitution can never offset the costs incurred from its green investments. Additionally, the CM, in an attempt to gain more excess demand, will raise the wholesale price to suppress the OEM's production. However, a sufficiently high wholesale price leads to a significant reduction in the OEM's production, resulting in a substantial decrease in the CM's profit from selling raw materials. Therefore, $\pi^{C(1,1)}$ is not optimal. In the (r, 0) scenario, the OEM can meet the CM's excess demand, leading the OEM to increase its production. In this case, the CM gains more revenue by selling more raw materials, making $\pi^{C(r,0)}$ optimal. Additionally, this result indicates that under the co-opetitive mode, product substitution can potentially lead to better supply chain profits.

Next, we compare the corresponding profits of the two modes under the (0, r) scenario. Firstly, from Figure 11, we observe that regardless of the values of k_o and k_c , $\pi_o^{P(0,r)} > \pi_o^{C(0,r)}$ always exists. This is because in the co-opetitive mode, CM acts as the supplier to OEM. In this mode, CM increases wholesale prices to suppress OEM's production to gain more demand. Consequently, OEM's yields decrease, leading to reduced sales revenue and ultimately lower profits.



Figure 11. Comparison of π_o under the (0, r) scenario. (a) $k_c = 3.5$. (b) $k_c = 6.5$.

From Figure 12, we observe that $\pi_c^{C(0,r)}$ always being less than $\pi_c^{P(0,r)}$. This is because under the co-opetitive mode, CM increases wholesale prices to suppress OEM production to obtain more excess demand. However, excessively high wholesale prices lead to a significant reduction in CM's profits from selling raw materials due to the substantial reduction in OEM production. At the same time, the excess demand obtained by CM cannot bring in a sufficiently large profit, leading to $\pi_c^{C(0,r)}$ always being less than $\pi_c^{P(0,r)}$.



Figure 12. Comparison of π_c under the (0, r) scenario. (a) $k_c = 3.5$. (b) $k_c = 6.5$.

Similarly to the results from Figures 11 and 12, we observe from Figure 13 that $\pi^{C(0,r)} < \pi^{P(0,r)}$. This is also due to the co-opetitive mode, where CM increases wholesale prices to suppress OEM production to obtain more excess demand. However, excessively high wholesale prices lead to a significant reduction in OEM production, resulting in a substantial decrease in CM's profits from selling raw materials. Additionally, the excess demand obtained by CM cannot bring in a sufficiently large profit, resulting in $\pi^{C(0,r)} < \pi^{P(0,r)}$. This result indicates that when only CM can substitute for OEM, the co-opetitive mode may not bring better profits to the supply chain.



Figure 13. Comparison of π under the (0, r) scenario. (a) $k_c = 3.5$. (b) $k_c = 6.5$.

Finally, we compare the respective profits of the two modes under the (1, 1) scenario. Firstly, Figure 14 illustrates that with the increase in k_o , $\pi_o^{C(1,1)} > \pi_o^{P(1,1)}$ emerges. This is because, in this scenario, the products of CM and OEM can substitute for each other. At this point, as k_o increases, OEM can fulfill more excess demand that CM cannot satisfy. In the co-opetitive mode, CM, acting as a supplier to OEM, offers lower wholesale prices to stimulate OEM to increase orders to gain more profit. Consequently, OEM can obtain more sales revenue at a lower ordering cost, leading to $\pi_o^{C(1,1)} > \pi_o^{P(1,1)}$.



Figure 14. Comparison of π_o under the (1, 1) scenario. (a) $k_c = 3.5$. (b) $k_c = 6.5$.

In Figure 15, we observe that when k_c is small, with the increase in k_o , we will sequentially encounter $\pi_c^{C(1,1)} < \pi_c^{P(1,1)}, \pi_c^{C(1,1)} > \pi_c^{P(1,1)}$ and $\pi_c^{C(1,1)} < \pi_c^{P(1,1)}$. However, when k_c is large, only when k_o is small, $\pi_c^{C(1,1)} > \pi_c^{P(1,1)}$ appears. This is because when k_c and k_0 are small, in the co-opetitive mode, CM increases the wholesale price to expand its substitutable demand. However, the significant increase in the wholesale price results in a substantial reduction in OEM's production, leading to greater losses, thus resulting in $\pi_c^{C(1,1)} < \pi_c^{P(1,1)}$. As k_o increases, the excess demand that OEM cannot satisfy decreases. In this scenario, CM increases the wholesale price to boost its additional demand for sales revenue. At this point, the extra sales revenue obtained by CM far exceeds the loss caused by the increase in the wholesale price, leading to $\pi_c^{C(1,1)} > \pi_c^{P(1,1)}$. Nevertheless, as k_o continues to increase, the excess demand that OEM cannot satisfy decreases significantly. At this point, CM reduces the wholesale price to stimulate OEM to further increase production. However, the loss caused by the decrease in the wholesale price is sufficiently significant, resulting in $\pi_c^{C(1,1)} < \pi_c^{P(1,1)}$. When k_c is large, CM's demand that can be satisfied is substantial. If k_0 is small, CM in the co-opetitive mode increases the wholesale price to obtain more substitutable demand. At this point, CM in the co-opetitive mode can obtain more sales revenue, resulting in $\pi_c^{C(1,1)} > \pi_c^{P(1,1)}$. However, as k_o increases, CM in the co-opetitive mode increases the wholesale price without obtaining more substitutable demand, but instead significantly reduces its revenue from selling raw materials. Therefore, $\pi_{c}^{C(1,1)} < \pi_{c}^{P(1,1)}$ occurs.



Figure 15. Comparison of π_c under the (1, 1) scenario. (a) $k_c = 3.5$. (b) $k_c = 6.5$.

In Figure 16, we observe that when k_c is small, $\pi^{C(1,1)} < \pi^{P(1,1)}$ always exists. When k_c is large, $\pi^{C(1,1)} > \pi^{P(1,1)}$ appears. This is because when k_c is small, in the co-opetitive mode, CM seeks to maximize its profits by manipulating the wholesale price. Although in

certain situations, manipulating the wholesale price can bring more profits to CM, it also causes greater damage to OEM's profits; hence, $\pi^{C(1,1)} < \pi^{P(1,1)}$ always exists. When k_c is large, in the co-opetitive mode, although CM seeks to maximize its profits by manipulating the wholesale price, this operation may cause less damage to OEM's profits. Therefore, $\pi^{C(1,1)} > \pi^{P(1,1)}$ appears. This result indicates that the co-opetitive mode may also bring more profits to the supply chain.



Figure 16. Comparison of π under the (1, 1) scenario. (a) $k_c = 3.5$. (b) $k_c = 6.5$.

7. Discussion and Conclusions

This study investigates the competitive newsvendor problem with product substitution under the carbon cap-and-trade system. Unlike studies such as those by refs. [23,43], our research introduces a carbon cap-and-trade system under conditions of market demand uncertainty. Additionally, we consider the scenario where the OEM has the option to procure raw materials from the CM or from a third-party supplier TS, and both of their final products are substitutable. Both OEM and CM products result in carbon emissions, which are offset through the carbon cap-and-trade system. Based on this problem, we consider different substitution relationships between OEM and CM products under both purely competitive and co-opetitive modes. Subsequently, we develop decision models for different scenarios and provide corresponding optimal solutions under these conditions. This approach enhances our understanding of the competitive issues when product substitution is considered.

Based on the obtained optimal solutions, we first analyze the impact of substitution rates, green investments, and carbon trading prices on the optimal decisions of OEM and CM under both purely competitive and co-opetitive modes. Unlike studies such as refs. [31,35] which did not consider product substitution and market demand uncertainty, we find some new findings based on the consideration of these factors. In purely competitive mode, when only one-way substitution exists, an increase in the substitution rate and green investment of OEM (or CM) leads to an increase in OEM's (or CM's) yields, while an increase in the green investment of OEM (or CM) does not necessarily reduce yields of CM (or OEM). In co-opetitive mode, when only one-way substitution exists, an increase in the substitution rate and green investments of both manufactures may lead to an increase in yields of both manufactures. Moreover, when only one-way substitution exists, an increase in carbon trading prices in the purely competitive mode may only lead to an increase in yields of the substitutable manufacturer, while an increase in carbon trading prices in the co-opetitive mode may lead to an increase in yields of both manufactures. Next, by comparing the two modes under the same substitution conditions, we find that when CM cannot substitute for OEM, the optimal decisions and total profits of the supply chain are the same in both modes. Finally, through numerical analysis, we further discovered that under the same competitive mode, under specific conditions, when the products of the two manufacturers can substitute for each other, it can bring better profits to both manufacturers and the entire supply chain. In contrast to the findings of ref. [11], we found

that under different competitive modes, when CM can substitute for OEM, neither mode is necessarily optimal for the profits of both manufacturers and the total profit of the supply chain. Furthermore, by comparing different substitution scenarios within the same mode, we found that under certain conditions, the existence of product substitution can enhance the total profit of the supply chain.

This study also provides new insights into operational decision-making for manufacturers in competitive environments. Firstly, under specific conditions, the existence of product substitution not only helps increase the production and profits of manufacturers but also enhances the total profit of the supply chain. Secondly, the study points out that under specific conditions, an increase in carbon trading prices can prompt manufacturers to increase their production. Therefore, manufacturers in competitive environments can alleviate the inhibitory effect of carbon emissions reduction on their production by participating in carbon cap-and-trade systems in specific environments. Furthermore, when only OEM products can substitute for CM products, whether or not the two manufacturers co-operate does not affect the total profit of the supply chain. Lastly, when competing manufacturers participate in the carbon cap-and-trade system, under specific conditions, the blind increase in wholesale prices by CM in the co-opetitive mode to seize the market will lead to a lose-lose situation. Therefore, CM in the co-opetitive mode should not blindly increase wholesale prices to seize the market.

The present study still has certain limitations. Firstly, it assumes that the green investments of OEM and CM are predetermined. However, in the actual operation of the supply chain, the green investments of OEM and CM can influence the optimal decisions of supply chain members. Therefore, treating the green investments of OEM and CM as decision variables for both parties simultaneously could pose an interesting new research question. Additionally, the study overlooks the flexibility of the carbon cap-and-trade system, as well as the uncertainty of carbon trading prices. It only assumes that OEM and CM settle their carbon quotas through the carbon cap-and-trade system at the end of the period. Hence, exploring carbon trading by OEM and CM throughout the period could be another intriguing research avenue.

Supplementary Materials: The following supporting information can be downloaded at: https://www.mdpi.com/article/10.3390/systems12060201/s1, Supplementary Materials: All Proofs.

Author Contributions: Conceptualization, Y.R.; methodology, Y.R.; software, Y.R. and H.L.; writing original draft preparation, Y.R.; writing—review and editing, H.L. and W.B.; visualization, Y.R. and Y.J.; project administration and funding acquisition, W.B. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Fundamental Research Funds for the Central Universities, grant number 2022JBW8005 and the Beijing Social Science Foundation (No. 22GLB035).

Data Availability Statement: The data presented in this study are available on request from the author.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- 1. Saen, R.F.; Moghaddas, Z.; Azadi, M. An advanced data analytic approach for reallocating green gas emissions in cap-and-trade context. *Ann. Oper. Res.* **2024**. [CrossRef]
- Chen, Z.-r.; Nie, P.-y. Implications of a cap-and-trade system for emission reductions under an asymmetric duopoly. *Bus. Strategy Environ.* 2024, 29, 3135–3145. [CrossRef]
- 3. Dormady, N.C. Market power in cap-and-trade auctions: A Monte Carlo approach. Energy Policy 2013, 62, 788–797. [CrossRef]
- Feng, H.; Zeng, Y.; Cai, X.; Qian, Q.; Zhou, Y. Altruistic profit allocation rules for joint replenishment with carbon cap-and-trade policy. *Eur. J. Oper. Res.* 2021, 290, 956–967. [CrossRef]
- Chen, Y.-h.; Wang, C.; Nie, P.-y.; Chen, Z.-r. A clean innovation comparison between carbon tax and cap-and-trade system. *Energy* Strategy Rev. 2020, 29, 100483. [CrossRef]
- Cai, J.; Jiang, F. Decision models of pricing and carbon emission reduction for low-carbon supply chain under cap-and-trade regulation. *Int. J. Prod. Econ.* 2023, 264, 108964. [CrossRef]

- Gao, D.; Tan, L.; Mo, X.; Xiong, R. Blue Sky Defense for Carbon Emission Trading Policies: A Perspective on the Spatial Spillover Effects of Total Factor Carbon Efficiency. Systems 2023, 11, 382. [CrossRef]
- 8. Du, S.; Zhu, Y.; Zhu, Y.; Tang, W. Allocation policy considering firm's time-varying emission reduction in a cap-and-trade system. *Ann. Oper. Res.* **2020**, *290*, 543–565. [CrossRef]
- 9. Wang, J.; Wan, Q.; Yu, M. Green supply chain network design considering chain-to-chain competition on price and carbon emission. *Comput. Ind. Eng.* 2020, 145, 106503. [CrossRef]
- 10. Xue, K.; Sun, G. Impacts of Supply Chain Competition on Firms' Carbon Emission Reduction and Social Welfare under Cap-and-Trade Regulation. *Int. J. Environ. Res. Public Health* **2022**, *19*, 3226. [CrossRef]
- 11. Dong, B.; Ren, Y.; McIntosh, C. A co-opetitive newsvendor model with product substitution and a wholesale price contract. *Eur. J. Oper. Res.* **2023**, *311*, 502–514. [CrossRef]
- 12. Chaudhuri, A.; Holbrook, M.B. The Chain of Effects from Brand Trust and Brand Affect to Brand Performance: The Role of Brand Loyalty. *J. Mark.* 2001, *65*, 81–93. [CrossRef]
- 13. Yang, H.; Luo, J.; Zhang, Q. Supplier encroachment under nonlinear pricing with imperfect substitutes: Bargaining power versus revenue-sharing. *Eur. J. Oper. Res.* 2018, 267, 1089–1101. [CrossRef]
- 14. Luo, Z.; Chen, X.; Wang, X. The role of co-opetition in low carbon manufacturing. Eur. J. Oper. Res. 2016, 253, 392–403. [CrossRef]
- 15. Dye, C.-Y.; Yang, C.-T. Sustainable trade credit and replenishment decisions with credit-linked demand under carbon emission constraints. *Eur. J. Oper. Res.* 2015, 244, 187–200. [CrossRef]
- Ma, J.; Liu, Y.; Geng, L. Does green credit financing mode with cap-and-trade scheme really benefit all members? *Manag. Decis. Econ.* 2023, 44, 4173–4192. [CrossRef]
- 17. Mandal, P.; Basu, P.; Choi, T.-M.; Rath, S.B. Platform financing vs. bank financing: Strategic choice of financing mode under seller competition. *Eur. J. Oper. Res.* 2023. [CrossRef]
- Wang, M.; Zhao, L.; Herty, M. Joint replenishment and carbon trading in fresh food supply chains. *Eur. J. Oper. Res.* 2019, 277, 561–573. [CrossRef]
- 19. Taleizadeh, A.A.; Shahriari, M.; Sana, S.S. Pricing and Coordination Strategies in a Dual Channel Supply Chain with Green Production under Cap and Trade Regulation. *Sustainability* **2021**, *13*, 12232. [CrossRef]
- 20. Ren, Y.; Bian, W.; Li, H.; Ma, X. Ordering Decisions with an Unreliable Supplier under the Carbon Cap-and-Trade System. *Sustainability* **2023**, *15*, 16595. [CrossRef]
- Yang, D.; Li, J.; Wang, B.; Jia, Y.-j. Module-Based Product Configuration Decisions Considering Both Economical and Carbon Emission -Related Environmental Factors. *Sustainability* 2020, 12, 1145. [CrossRef]
- Li, C.; Wang, J.; Shi, Y. The Impact of Government Intervention and Cap-and-Trade on Green Innovation in Supply Chains: A Social Welfare Perspective. Sustainability 2022, 14, 7941. [CrossRef]
- 23. Niu, B.; Chen, Y.; Zeng, F. One step further for procurement cooperation: Will the industry leader benefit from its competitive manufacturer's joint determination of consumption quality? *Eur. J. Oper. Res.* **2023**, *311*, 989–1008. [CrossRef]
- Jiang, W.; Yuan, M. Coordination of Prefabricated Construction Supply Chain under Cap-and-Trade Policy Considering Consumer Environmental Awareness. Sustainability 2022, 14, 5724. [CrossRef]
- 25. Marufuzzaman, M.; Eksioglu, S.D.; Hernandez, R. Environmentally Friendly Supply Chain Planning and Design for Biodiesel Production via Wastewater Sludge. *Transp. Sci.* **2014**, *48*, 555–574. [CrossRef]
- 26. Drake, D.F.; Kleindorfer, P.R.; Van Wassenhove, L.N. Technology Choice and Capacity Portfolios under Emissions Regulation. *Prod. Oper. Manag.* 2016, 25, 1006–1025. [CrossRef]
- 27. Zhang, B.; Xu, L. Multi-item production planning with carbon cap and trade mechanism. *Int. J. Prod. Econ.* **2013**, 144, 118–127. [CrossRef]
- Qin, J.; Bai, X.; Xia, L. Sustainable Trade Credit and Replenishment Policies under the Cap-And-Trade and Carbon Tax Regulations. Sustainability 2015, 7, 16340–16361. [CrossRef]
- 29. Yang, W.; Ma, J.; Fan, A.; Zhang, J.; Pan, Y. Effectiveness of policies for electric commercial vehicle adoption and emission reduction in the logistics industry. *Energy Policy* **2024**, *188*, 114116. [CrossRef]
- 30. Ferguson, M.E.; Toktay, L.B. The Effect of Competition on Recovery Strategies. Prod. Oper. Manag. 2006, 15, 351–368. [CrossRef]
- 31. Chai, Q.; Xiao, Z.; Zhou, G. Competitive strategies for original equipment manufacturers considering carbon cap and trade. *Transp. Res. Part Transp. Environ.* **2020**, *78*, 102193. [CrossRef]
- 32. Bolandifar, E.; Chen, Z.; Zhu, K. Managing competitive levers in a collaborative distribution channel. *Eur. J. Oper. Res.* **2021**, 293, 1031–1042. [CrossRef]
- 33. Kim, Y. Optimal Timing Strategies in the Evolutionary Dynamics of Competitive Supply Chains. Systems 2024, 12, 114. [CrossRef]
- 34. Devine, M.T.; Siddiqui, S. Strategic investment decisions in an oligopoly with a competitive fringe: An equilibrium problem with equilibrium constraints approach. *Eur. J. Oper. Res.* **2023**, *306*, 1473–1494. [CrossRef]
- 35. Luo, H.; Niu, B. mpact of competition type on a competitive manufacturer's preference of decision timing. *Int. J. Prod. Econ.* **2022**, 251, 108548. [CrossRef]
- 36. Chen, Y.; Bai, Q.; Xu, J. Comparison of competing supply chains with different structures under cap-and-trade regulation. *Rairo-Oper. Res.* **2024**, *58*, 1653–1680. [CrossRef]
- 37. Du, S.; Huang, C.; Yan, X.; Tang, W. Voluntary green technology adoption: The effects of regulatory uncertainty and competition. *Eur. J. Oper. Res.* **2024**, *315*, 528–540. [CrossRef]

- 38. Shan, X.; Xiong, S.; Zhang, C. Mitigating supply disruption risks by diversifying competing suppliers and using sales effort. *Int. J. Prod. Econ.* **2023**, 255, 108637. [CrossRef]
- 39. Straubert, C.; Sucky, E. Inventory competition on electronic marketplaces—A competitive newsvendor problem with a unilateral sales commission fee. *Eur. J. Oper. Res.* 2023, *309*, 656–670. [CrossRef]
- 40. Lippman, S.A.; McCardle, K. F. The competitive newsboy. Oper. Res. 1997, 45, 54–65. [CrossRef]
- 41. Niu, B.; Xie, F.; Chen, L. Analysis of global brands' green procurement strategy: Co-opetitive local sourcing vs. overseas sourcing under delivery disruption. *Omega* **2024**, *123*, 102991. [CrossRef]
- 42. Anand, K.S.; Giraud-Carrier, F.C. Pollution Regulation of Competitive Markets. Manag. Sci. 2020, 66, 4193-4206. [CrossRef]
- 43. Niu, B.; Zhang, N.; Zhang, J. How to Induce Multinational Firms' Local Sourcing to Break Carbon Lock-in? *Eur. J. Oper. Res.* 2023, 315, 613–625. [CrossRef]
- 44. Zhu, Y.; Zhang, H.; Siddik, A.B.; Zheng, Y.; Sobhani, F.A. Understanding Corporate Green Competitive Advantage through Green Technology Adoption and Green Dynamic Capabilities: Does Green Product Innovation Matter? *Systems* 2023, *11*, 461. [CrossRef]
- Rehman, S.U.; Chan, M.P.; Almakhayitah, M.Y.; Albakhit, A.I.A.; Abdou, A.H. Going green! Factors influencing green competitive advantage of Chinese SMEs: A moderated-mediated perspective. *Environ. Sci. Pollut. Res.* 2024, 31, 15302–15320. [CrossRef] [PubMed]
- Liu, G.; Chen, J.; Li, Z. Green financing strategies under risk aversion and manufacturer competition. *Rairo-Oper. Res.* 2024, 58, 1927–1954. [CrossRef]
- 47. Huang, J.; Huang, H.; Si, Y.; Xu, Y.; Liu, S.; Yang, X. Green supply chain finance strategies with market competition and financial constraints. *Heliyon* **2024**, *10*, e29511. [CrossRef]
- 48. An, S.; Li, B.; Song, D.; Chen, X. Green credit financing versus trade credit financing in a supply chain with carbon emission limits. *Eur. J. Oper. Res.* 2021, 292, 125–142. [CrossRef]
- 49. Lariviere, M.A.; Porteus, E.L. Selling to the Newsvendor: An Analysis of Price-Only Contracts. *Manuf. Serv. Oper. Manag.* 2001, *3*, 293–305. [CrossRef]
- 50. Jing, B.; Chen, X.; Cai, G. Equilibrium Financing in a Distribution Channel with Capital Constraint. *Prod. Oper. Manag.* 2012, 21, 1090–1101. [CrossRef]
- 51. Ovchinnikov, A.; Moritz, B.; Quiroga, B. F. How to Compete Against a Behavioral Newsvendor. *Prod. Oper. Manag.* 2015, 24, 1783–1793. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.